# DIGITALES ARCHIV

ZBW – Leibniz-Informationszentrum Wirtschaft ZBW – Leibniz Information Centre for Economics

Shapovalenko, Nadiia; Vdovychenko, Artem

# Article Interest rate pass-through in Ukraine : estimates and determinants

Visnyk Nacional'noho Banku Ukraïny

**Provided in Cooperation with:** ZBW OAS

*Reference:* Shapovalenko, Nadiia/Vdovychenko, Artem (2023). Interest rate pass-through in Ukraine : estimates and determinants. In: Visnyk Nacional'noho Banku Ukraïny (255), S. 22 - 70. https://journal.bank.gov.ua/download/article/2023/255/02/en. doi:10.26531/vnbu2023.255.02.

This Version is available at: http://hdl.handle.net/11159/703158

**Kontakt/Contact** ZBW – Leibniz-Informationszentrum Wirtschaft/Leibniz Information Centre for Economics Düsternbrooker Weg 120 24105 Kiel (Germany) E-Mail: *rights[at]zbw.eu* https://www.zbw.eu/

### Standard-Nutzungsbedingungen:

Dieses Dokument darf zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden. Sie dürfen dieses Dokument nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, aufführen, vertreiben oder anderweitig nutzen. Sofern für das Dokument eine Open-Content-Lizenz verwendet wurde, so gelten abweichend von diesen Nutzungsbedingungen die in der Lizenz gewährten Nutzungsrechte. Alle auf diesem Vorblatt angegebenen Informationen einschließlich der Rechteinformationen (z.B. Nennung einer Creative Commons Lizenz) wurden automatisch generiert und müssen durch Nutzer:innen vor einer Nachnutzung sorgfältig überprüft werden. Die Lizenzangaben stammen aus Publikationsmetadaten und können Fehler oder Ungenauigkeiten enthalten.



BY NC https://savearchive.zbw.eu/termsofuse

### Terms of use:

This document may be saved and copied for your personal and scholarly purposes. You are not to copy it for public or commercial purposes, to exhibit the document in public, to perform, distribute or otherwise use the document in public. If the document is made available under a Creative Commons Licence you may exercise further usage rights as specified in the licence. All information provided on this publication cover sheet, including copyright details (e.g. indication of a Creative Commons license), was automatically generated and must be carefully reviewed by users prior to reuse. The license information is derived from publication metadata and may contain errors or inaccuracies.



ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

# INTEREST RATE PASS-THROUGH IN UKRAINE: ESTIMATES AND DETERMINANTS

# NADIIA SHAPOVALENKO<sup>a</sup>, ARTEM VDOVYCHENKO<sup>a</sup>

<sup>a</sup>National Bank of Ukraine

E-mail: nadiia.shapovalenko@bank.gov.ua artem.vdovychenko@bank.gov.ua

Abstract

ct In this study, we apply ARDL models to estimate the strength of long-run interest rate pass-through in Ukraine. We focus on the transmission of the overnight interbank interest rate to the rates on term deposits of households and loans to non-financial corporations – both in national currency. Controlling for macroeconomic indicators and bank financial variables we obtain bank-level time-varying estimates of transmission and run a set of panel regressions to analyze the determinants of pass-through strength. Besides linear estimates, we report asymmetric transmissions, which differ depending on the decrease or increase in the interbank rate, and timevarying estimates for transmission.

**JEL Codes** E43, E52, E58

**Keywords** interest rate transmission, determinants of transmission, ARDL, panel data

**Acknowledgments:** We are grateful to Yuliia Kuzmenko and Mykhailo Rebryk for comments and suggestions. We also thank Nicolas Aragon, Pervin Dadashova, Petro Dovhyi, and Maksym Drobiazhin for their help in data gathering and method development.

# **1. INTRODUCTION**

The strength of interest rate pass-through<sup>1</sup> is an important characteristic of monetary policy efficiency. Deposit rates<sup>2</sup> affect households' decisions on consumption, savings and their demand for foreign currencies. Credit rates<sup>3</sup> determine business activity and the population's propensity to consume, driving aggregate demand in an economy. These decisions significantly influence GDP, exchange rates, and inflation dynamics, which in turn demonstrate how effectively a central bank is achieving its goals. In 2016 the National Bank of Ukraine moved to an inflation targeting regime. Under this framework, the interest rate channel plays an important role in monetary transmission. Therefore, estimating the strength of pass-through and understanding its functioning becomes an essential issue.

Understanding transmission is even more important from the monetary policy perspective if there is a switch to a hybrid monetary regime, as happened in Ukraine after the full-scale military invasion of russia. This regime was introduced in the first days of the invasion and it assumes a smooth transition from a fixed exchange rate back to inflation targeting. The new goal of monetary policy in such a period of transition

is to support "flexible exchange rate sustainability". To achieve this goal, the NBU sets its interest rate at a level that makes assets denominated in national currency attractive to households. Following this logic, monetary decisions help to control both - inflation and demand for foreign currency. These circumstances increase the motivation to study transmission, especially in a nonlinear framework. The central bank needs to know whether the reaction of deposit rates is symmetrical for a decrease/increase of the key rate. If deposit rates are more responsive to key rate decreases, then the monetary authorities must be careful in loosening monetary policy as the attractiveness of assets in domestic currency can be lost very fast. Another important issue is the reaction of the banks regarding to their size and ownership status. Under these conditions, many questions can be asked regarding transmission estimates.

In this study, we estimate the strength of transmission from changes in the overnight interbank rate (IR) to deposit and credit rates in Ukraine over a range of horizons. Our estimates are based on the application of ARDL models. Extending previous studies on interest rate transmission for Ukraine, we apply nonlinear regressions that allow us to estimate asymmetry in transmission and its change over

<sup>&</sup>lt;sup>1</sup> In this paper, we use the concepts of interest rate pass-through and transmission of interbank interest rate to banks' rates interchangebly.

<sup>&</sup>lt;sup>2</sup> Hereafter under deposits, we mean term deposits of households (HH) in national currency (hryvnias).

<sup>&</sup>lt;sup>3</sup> Hereafter under credits, we mean loans to non-financial corporations (NFC) in national currency (hryvnias)

<sup>©</sup> National Bank of Ukraine, N. Shapovalenko, A. Vdovychenko, 2023. This work is licensed under a Creative Commons Attribution-NonCommercial 4.0 International License. Available at https://doi.org/10.26531/vnbu2023.255.02.

time. We estimate transmission strength and its dynamics both at the level of the whole banking system and at the level of individual banks. Estimates of time-varying transmission allow us to run panel regression to detect the determinants of pass-through at the macroeconomic and bank levels. The results are useful to the monetary authorities, as we demonstrate recent tendencies in the dynamics of passthrough strength and analyze the characteristics of banks that affect pass-through.

The rest of this paper is organized as follows: Section 2 presents a review of the literature on transmission estimates and the analysis of its determinants. Section 3 presents the methodology for estimating transmission strength and discusses the variables we use as potential determinants. Section 4 provides data characteristics. Section 5 reports the main results of our estimates for Ukraine. Conclusions are presented in Section 6.

### **2. LITERATURE REVIEW**

Most studies of interest rate pass-through show that the reaction of market rates is sluggish and incomplete over the long run. Incomplete transmission is seen especially in deposit rates, which are included in banks' expenses and for this reason are rigid to increase. In theory, pass-through would be complete in an environment of full information, perfect competition, and risk-neutral banks (Gigineishvili, 2011). In the literature, imperfect transmission is usually explained by a low level of competition between banks, the availability of alternative sources of funding, the rigidity of banks' costs, or the high elasticity of consumer demand with respect to interest rate adjustments. Additional reasons are the possible deterioration of the creditworthiness of borrowers, implicit interest rate fixing for clients involved in long-term relationships with banks, the structure of bank rates where long-term contracts can dominate, and uncertainty over the future dynamics of money market rates (Mojon, 2000). However, there are also arguments for longrun transmission over unity, mostly applicable to credit rates. Under high credit risks, banks may not ration credits but increase the risk premium for loans instead (Bondt, 2005). This "overshooting" reflects the asymmetry of information between banks and borrowers (Sørensen and Werner, 2006).

Interest rate transmission is also found to be asymmetric in different aspects, while asymmetry itself is often the object of empirical studies. For instance, pass-through to credit rates is higher in the periods when money market rates (MMRs) are increasing than when they are decreasing (Mojon, 2000). Asymmetry works in the opposite direction for deposit rates, which are less rigid for decrease, suggesting that banks hold some degree of pricing power in the markets. Sander and Kleimeier (2002, 2004) demonstrate that asymmetries exist not only in the direction of rate changes but also in the severity of the interest rate shock. This "threshold" reaction means that banks take into account menu costs and transmit changes in market rates of a minimum size. Aristei and Gallo (2014) use a Markov-switching vector autoregressive model to show that the interest rate pass-through during periods of financial distress is lower than in normal times.

Cross-country studies found a large amount of variability in interest rate transmission strength between economies. Based on a literature survey, Mishra and Montel (2012) conclude that monetary policy transmission is much weaker in developing countries than in advanced economies.

This typical conclusion from empirical studies initiated an important strand of the literature devoted to the differences in the strength of transmission. In a pioneering paper, Cottarelli and Kourelis (1994) examine lending rate stickiness across a panel of countries. They attribute the terminated and incomplete response of lending rates to MMR changes to financial structure, certain characteristics of which create frictions that prevent full adjustment. Among such characteristics, the authors identify factors affecting demand for loans and the ability of banks to change rates quickly: the level of competition between banks, barriers to market entry, alternatives to bank loans, and administratively set discount rates. Adjustment costs and uncertainty also slow down the lending rate response, as banks can perceive MMR changes as only temporary. Cottarelli and Kourelis (1994) estimate the degree of lending rate stickiness for 31 industrial and developing countries by regressing respective rates against distributed lags of the money market and discount rates in each country. Stickiness is measured as the degree of response of lending rates to changes in MMRs at different time lags. In the second stage, they regress a cross-section of stickiness measures against a set of country-specific financial system features. Specifically, it is shown that the variation in lending rate stickiness is explained by the existence of a sizable market for shortterm monetary instruments, the degree of capital mobility, constraints on bank competition, private sector ownership of the banking system, the volatility of MMRs, and the level of inflation. A two-stage empirical procedure, in which passthrough measures are first estimated, and then regressed on a set of determinants, has become the standard approach in recent decades. For instance, Mojon (2000) analyzes the heterogeneity of transmission from MMRs to retail bank rates (25 credit rates and 17 deposit rates) for the six largest countries in the euro area (Belgium, Germany, Spain, France, Italy, and the Netherlands). To estimate the pass-through effect, he runs an error correction model with distributed lags for each kind of retail rate where the MMR is the only explanatory variable. In the second stage, the author builds a panel of retail bank markets and run panel regressions accounting for the heterogeneity of transmission measures across countries and over time. Analysis of the determinants of the pass-through focuses on four sets of variables: the monetary policy regime, competition among banks, competition from non-banking financial sectors, and the rigidity of bank costs. The fourth group of factors is particularly interesting, given that the rigidity of the costs is not frequently analyzed in studies on this topic. Mojon (2000) assumes that a higher share of operating costs in the total costs of a bank should imply a smaller pass-through. Unlike funding costs, operating costs are weakly related to the movements of MMRs. It is also argued that a bank that relies on traditional deposits is more likely to have more rigid funding costs (meaning lower pass-through to credit rates) than a bank that prefers debt as a source of funding. Their empirical evidence shows that the structure of bank costs matters. Specifically, higher staff costs result in a smaller degree of pass-through to credit rates.

Studies of IR transmission to bank rates have been carried out earlier in Ukraine. One paper, by Hlazunov et al (2023), is focused on the transmission of the key policy rate to deposit and credit rates depending on the banking sector's structure. The authors differentiate their estimates based on loan and deposit maturities, as well as the ownership status of the banks. Assuming that a bank's efficiency depends on its type of ownership, they conclude that less efficient banks have more powerful transmission to deposit rates and weaker pass-through to credit rates. The interest rate pass-through is lower for deposits and loans with higher maturities.

From the perspective of an analysis of the monetary transmission mechanism in Ukraine, it is worth discussing whether policy tools (such as reserve requirements) can be effectively used to enhance transmission. The literature on this topic is rather scant, and conclusions differ depending on the specific tool chosen. For instance, Ma and Wang (2014) based on the results from a DSGE model for China, found that institutional constraints (such as a high deposit reserve requirement and quantitative limits on the loanto-deposit ratio or caps on loans) increase the cost of rebalancing assets and liabilities, or directly limit asset allocation. These restrictions thus weaken the transmission of policy rates to deposit and lending rates. However, Terrier et al (2011) argue that reserve requirements can be a useful instrument to bring the interbank rate close to the policy rate under conditions of excessive liquidity or stress in the financial system. Generally speaking, the variables used in empirical literature to explain heterogeneity in monetary policy transmission across countries or individual banks can be grouped into macroeconomic factors, financial structure characteristics, and the balance sheet characteristics of individual banks (Sørensen and Werner, 2006; Leuvensteijn et al, 2008; Gigineishvili, 2011; Saborowski and Weber, 2013; Stanisławska, 2015; Leroy and Lucotte, 2016).

In studies on interest rate transmission, the standard set of macroeconomic determinants usually includes inflation, the exchange rate regime, and the business cycle. The financial structure of an economy affects transmission through competition in the banking system, the development of the nonbanking sector, the share of state-owned or foreign banks, the level of non-performing loans, and the level of liquidity in the banking system. The returns on equity and assets, net interest margin, cost-to-income ratio, cost structure, liquidity ratio, capital adequacy ratio, funding structure, and credit portfolio quality belong to control determinants taken from banks' balance sheets.<sup>4</sup>

A comprehensive review of econometric techniques used to estimate interest rate transmission strength and its determinants is presented in Deutsche Bundesbank (2019). The standard approach is to apply some modification of the error correction model (ECM) to time series or panel data to obtain long-run responses of the banks' rates to the MMR. ECM is usually estimated in the form of ARDL, via dynamic OLS (DOLS) or dynamic seemingly unrelated regression (DSUR) approaches (Mojon, 2000; Sørensen and Werner, 2006; Leuvensteijn et al, 2008; Gigineishvili, 2011; Leroy and Lucotte, 2016; Gregor and Melecký, 2018; Fičura and Witzany, 2023). A multivariate approach to the estimation of long-run relationships between interest rates is also popular. Saborowski and Weber (2013) argue that ARDL specifications are not appropriate, as there is feedback from the policy rate to movements in retail rates, meaning that the policy rate is not purely exogenous. Instead, they suggest using the VAR framework for estimates. Specifically, Saborowski and Weber (2013) estimate the panel VAR (PVAR) model, both with and without interaction terms (which are variables potentially affecting transmission), to determine the average pass-through across countries and identify determinants of interest rate transmission. Leroy and Lucotte (2016) also use

interaction terms to analyze their impact on transmission in both ARDL and VAR frameworks.

# 3. ECONOMETRIC APPROACH TO MEASURING THE PASS-THROUGH

### **3.1. Linear vs Nonlinear ARDL Models**

In recent years, the majority of studies have utilized error correction models (ECM) to evaluate interest rate pass-through. The ECM is usually estimated by ARDL models, which enable the examination of both the longterm equilibrium pass-through of MMRs to retail bank rates and the speed at which adjustments occur toward this equilibrium.

To estimate the transmission from MMR x to bank rate y, a standard ARDL(p,q) of a form (1) is reformulated into an error correction form denoted as (2):

$$y_{t} = c_{0} + \sum_{i=1}^{p} \beta_{y_{i}} y_{t-i} + \sum_{i=0}^{q} \beta_{x_{i}} x_{t-i} + \epsilon_{t},$$
(1)

$$\Delta y_{t} = c_{0} + \alpha \left( y_{t-1} - \theta x_{t-1} \right) + \sum_{i=1}^{p-1} \psi_{y_{i}} \Delta y_{t-i} + \sum_{i=0}^{q-1} \psi_{x_{i}} \Delta x_{t-i} + \epsilon_{t}, \quad (2)$$

where  $\theta$  is the value of long-term equilibrium pass-through and  $\alpha$  is the speed of correction.

In this study, we analyze  $\theta$ , which we interpret as a general characteristic of interest rate pass-through. Considering that  $\theta$  is an asymptotic value of transmission, we extended our analysis by estimating impulse response functions (IRF) for all of the transmissions we estimated. The IRFs show how banks' rates react to an MMR shock. This allows us to analyze pass-through strength at different horizons, disentangling short-, medium-, and long-run effects. At this stage, we ignore the speed of error correction ( $\alpha$ ), indicating how fast banks adjust their rates. This information can be useful for understanding transmission, especially for central banks, as it predicts the dynamics of banking rates in the short term. However, the speed of the banks' reaction is captured at least partially by our estimates of IRFs for different horizons. If a short-term IRF value covers, for example, more than half of the overall pass-through, this indicates a high speed of transmission.

To get information about average transmission values, we first estimate linear ARDLs for economy-level and bank-level data. ARDLs provide straightforward coefficient estimates, making it easier for policymakers and analysts to understand the relationships between the variables. We determine the optimal number of lags using the Bayesian Information Criterion (BIC), as it is commonly used to choose a more parsimonious model.

However, linear models fall short of addressing nonlinearity and asymmetry, which are commonly observed phenomena in economic data. This limitation can lead to misspecification errors, resulting in biased parameter estimates and inaccurate policy implications. Non-linear models offer greater flexibility in capturing complex relationships that cannot be adequately represented by linear models. This flexibility allows for a more accurate representation of economic phenomena with nonlinear patterns.

<sup>&</sup>lt;sup>4</sup> Overview of relevant studies and potential determinants of transmission are presented in more detail in Appendix A, Tables 2-3.

Specifically, we are interested in the asymmetries associated with the direction of change in the MMR. Commercial banks may exhibit different reactions when the central bank adjusts its interest rates upwards versus downwards. In instances of a central bank raising rates, commercial banks might be quicker to pass on the increased costs to borrowers, resulting in a relatively prompt upward adjustment in lending rates. Conversely, when the central bank lowers interest rates, commercial banks may display a more gradual response, with deposit rates adjusting at a slower pace. The literature exploring asymmetry includes instances where asymmetry is observed, as well as cases where it is not evident. See for example Gambacorta & lannotti (2007), Sznajderska (2012), Månsson et al. (2013), Apergis et al. (2015).

To verify whether there is an asymmetry in the interest rate pass-through for Ukraine, we apply the NARDL framework proposed by Shin et al. (2014). This approach captures shortterm and long-term nonlinearities by representing them as positive and negative partial sum decompositions of the distributed lag variables.

Noting, that any series  $z_t$  may be written as  $z_t = z^0 + z_t^+ + z_t^$ where  $z_t^-$  and  $z_t^+$  are the partial sum processes of positive and negative changes in  $z_t^-$ :

$$Z_t^+ = \sum_{s=1}^t \max(\Delta Z_s, \mathbf{0}), \qquad (3)$$

$$z_t^- = \sum_{s=1}^t \min(\Delta z_s, 0), \tag{4}$$

with some manipulation (2), may be rewritten as

$$\Delta y_{t} = c_{0} + \alpha \left( y_{t-1} - \left( \theta^{+} x_{t-1}^{+} + \theta^{-} x_{t-1}^{-} \right) \right) + \sum_{i=1}^{p-1} \psi_{y_{i}} \Delta y_{t-i} + \dots$$
$$\dots + \sum_{i=0}^{q-1} \left( \psi_{x_{i}}^{+} \Delta x_{t-i}^{+} + \psi_{x_{i}}^{-} \Delta x_{t-i}^{-} \right) + \epsilon_{t},$$
(5)

where asymmetric long-term equilibrium pass-through is given by the parameters  $\theta^{+}$  and  $\theta^{-}$ .

To verify the presence or absence of statistically significant asymmetry, a Wald test with the null hypothesis that  $\theta^{+} = \theta^{-}$  is conducted.

NARDL models are specifically designed to handle nonlinear relationships through their inclusion of positive and negative partial sum decompositions of distributed lag variables. This makes them well-suited for capturing asymmetries. However, they may not adequately address certain specific forms of asymmetry or nonlinearity. For example, abrupt changes or complex patterns in the datagenerating process might be challenging for traditional models to accommodate. In such cases, the introduction of additional flexibility, such as allowing the coefficients of the model to vary over time, as in models with timevarying parameters, becomes crucial. In the next step of our research, we apply the Bayesian approach to estimate ARDLs with time-varying parameters (TVP) to obtain a timevarying estimate of the long-run pass-through.

### **3.2. ARDL with Time-Varying Parameters**

ARDL models with time-varying parameters (TVP-ARDL) introduce an additional element of flexibility by allowing parameters to change over time. This adaptive approach

enables the model to better capture evolving relationships and structural shifts in the data, offering a more nuanced representation of the underlying dynamics.

Furthermore, having time-variant pass-through coefficients enables us to undergo regression analyses involving various sets of explanatory variables (i.e., determinants of pass-through). This process aims to pinpoint variables that demonstrate statistical significance, and meaningful relationships, and yield a satisfactory overall model fit.

The TVP-ARDL model we employ allows for time variation in both the model coefficients and the residual covariance matrix. We use Bayesian inference to estimate the parameters of the model.

For simplicity, consider a model in a form (1) with a timevarying parameters:

$$y_{t} = c_{t,0} + \sum_{i=1}^{p} \beta_{t,y_{i}} y_{t-i} + \sum_{i=0}^{q} \beta_{t,x_{i}} x_{t-i} + \epsilon_{t}, \quad \epsilon_{t} \sim N(0, \Sigma_{t}).$$
(6)

The model can be reformulated as

$$y_t = \overline{X_t} \omega_t + \epsilon_t, \tag{7}$$

where  $\overline{X_{t}} = (1y_{t-1} \ y_{t-2} \dots y_{t-p} \ x_{t} \ x_{t-1} \dots x_{t-q})$ 

$$\omega_t = \left( c_{t,0} \beta_{t,y_1} \beta_{t,y_2} \dots \beta_{t,y_p} y_{t-p} \beta_{t,x_1} \beta_{t,x_2} \dots \beta_{t,x_q} \right).$$

Coefficients in our model vary in time following a random walk process. Accordingly, the dynamic process for  $\omega_t$  is:

$$\omega_t = \omega_{t-1} + v_t, \ v_t \sim N(0, \Omega) \tag{8}$$

While  $\Sigma_t = \overline{s} \exp(\lambda_t)$  where  $\overline{s}$  is a known scaling term,  $\lambda_t$  is a dynamic process generating the heteroskedasticity of the model.

$$\lambda_t = \gamma \omega_{t-1} + \vartheta_t, \ \vartheta_t \sim N(0, \phi). \tag{9}$$

The parameters to estimate are coefficients  $\omega = \{\omega_t : t = 1,..,T\}$ , the covariance matrix  $\Omega$  for the shocks on the dynamic process, the set of dynamic coefficients  $\lambda = \{\lambda_t : t = 1,..,T\}$ , and the heteroskedasticity parameter  $\phi$ .

The Gibbs sampler algorithm is used for model parameter estimations. For a detailed algorithm, please refer to Dieppe et al. (2016).

The specifications of the estimated TVP-ARDLs align with those estimated in a previous section. Table 1 provides details regarding the estimation process and some parameters.

 Table 1. TVP-ARDL Parameters

Total number of iterations	50,000
Burn-in iterations	40,000
Post-burn selection	yes
Frequency of draw selection	20
AR coefficient on residual variance γ	0.85
IG shape on $\varphi \: \alpha_{_0}$	0.001
IG scale on $\varphi  \delta_{_0}$	0.001

### 3.3. Determinants of the Pass-Through

Time-varying estimates of transmission allow us to study potential determinants in a panel regression framework with banks as cross-section units and a monthly time domain. Running panel regressions, we include fixed effects to control for the specific features of separate banks and to avoid endogeneity caused by the omitted variables problem. We also control for time-specific shocks, applying two-way fixed effects. However, the latter is not always feasible, as economy-wide variables are invariant in the cross-section dimension, thus time-fixed effects suffer from collinearity. The panel regressions were estimated with different approaches to get standard errors of coefficient robust to clustering, heteroscedasticity, and the autocorrelation of residuals.

Potential determinants of IR pass-through to deposit and credit rates can be grouped into macroeconomic conditions, the structure of assets and liabilities, and the structure of revenues and expenditures (Appendix A).

### **Macroeconomic Variables**

The group of macroeconomic variables includes the level of liquidity in the banking system, the financial stress index (FSI), the Herfindahl-Hirschman index for deposits and credits, inflation, proxies for deposit supply, and credit demand.

We assume that *excessive liquidity* weakens transmission. It reduces incentives for banks to compete for depositors by raising deposit rates. For creditors, high liquidity can reflect a more conservative business model or a lack of investment opportunities.

Economic uncertainty, which we proxy by FSI (Filatov, 2021), potentially negatively affects transmission to deposit rates. Facing uncertainty, banks are less likely to change deposit rates sharply: the risk of deposit outflows prevents them from cutting rates, while there is a risk of a loss of profits from a significant increase in interest costs. For credit rates, the effect on transmission is not obvious. In a low-risk environment banks with weak balance sheets may react to an expansive monetary policy by shoring up liquidity rather than extending credit at lower rates. At the same time, the level of credit risk may also be related to the degree of competition. Even in periods of high uncertainty, banks in a more competitive environment would tend to lend to more risky borrowers to boost their market shares.

Banks competition in credit and deposit markets is quantified by the Herfindahl-Hirschman index. Market monopolization hinders transmission, as monopolists care little about the reaction of competitors and the possible loss of customers. However, this effect can be nonlinear in the sense that low competition weakens transmission to credit rates when the IR goes down but enforces pass-through when the rates increase (and vice versa for deposit rates).

The impact of *inflation* on the strength of transmission is explained, in part, by the need to index deposit rates. Higher inflation usually requires a more significant increase in interest rates to keep deposits attractive in real terms and make credit operations profitable. Lower inflation allows for lower rates without the threat of deposit outflows and losses from credits.

Deposits supply and credit demand are pro-cyclical. Economic growth and high credit demand encourage banks to allocate their funds towards riskier projects, meaning higher transmission on credit rates. However, periods of high demand are characterized by underestimated credit risk of borrowers and high competition for market share. These factors may lower real credit rates, decreasing transmission. When deposit supply is high it may be easier for banks to keep rates at their current level without losing clients, which means lower transmission. On the other hand, a high supply may be associated with a more dynamic market, leading to more entries and higher competition, thereby making deposit rates increase.

A separate subgroup of variables characterizes the actions of the NBU directed at enforcement of the transmission to deposit rates. The instruments recently used for these goals are the reserve requirement ratio for households' deposits on demand and current accounts, and the NBU's long-term certificates of deposit (CDs). Raising the reserve requirement ratio increases the attractiveness of term deposits for banks by reducing their opportunity cost. This encourages banks to attract term deposits, including by raising interest rates. The option of investing spare liquidity in longer-term CDs with higher yields also creates room for banks to raise deposit rates. Considering the potential importance of these instruments for transmission, we include them in regressions.

### **Balance Sheet Variables**

A high share of non-performing loans is expected to harm transmission to credit rates because banks with weak balance sheets prefer meeting liquidity requirements rather than extending credit portfolios. A reduction in the policy rate may thus have only a limited impact on market rates. The opposite effect can be detected in a highly competitive environment, when banks on average take on more problematic loans, which might explain the positive correlation with the pass-through. *Provisions to assets ratio* is another variable measuring the degree of risks for a bank. Similar to the share of non-performing loans, the level of reserves can have different effects on transmission depending on market concentration.

*Funding* cost must have a positive correlation with transmission to the deposit rates considering that households' deposits constitute a significant share of banking system liabilities. The impact on transmission to credit rates is questionable. If banks set credit rates by adding a margin over funding costs, then we can assume the positive effects of the cost increase on transmission.

Profit after taxation to equity capital ratio or net interest income to assets ratio (net interest margin) are interpreted as proxies for competition in the market. High profitability in the banking sector often weakens pass-through, since it reflects inadequate competition and market power. In an uncompetitive environment, banks can set higher premiums that deviate from their marginal costs. As a result, lending rates become less elastic due to changes in funding costs.

*Own capital to liabilities ratio*. A high capital ratio may act as a buffer against market fluctuations and would hence reduce the strength of transmission.

Short-term securities of the NBU to assets and government bonds to assets can have a positive effect on transmission to credit rates. This may happen because of the crowding-out effect, when banks allocate their resources to government securities and supply fewer credits to the economy. When the demand for credits goes up potential debtors have to compete for resources with the government. This allows the banks to set higher margins.

The deposits to liabilities ratio signals about stability of funding. A bank with a high share of traditional deposits in liabilities is more likely to have more stable funding costs than a bank that funds itself mostly by issuing debt on the capital markets. Potentially, banks with a large and stable share of deposit funding are expected to be less vulnerable to changes in MMRs, thereby leading to a relatively weaker transmission to credit rates.

#### The Structure of Revenues and Expenditures

Non-interest income share or the share of interest expenses reflect the diversification of money flows. A bank with a highly diversified portfolio of activities may be less sensitive to changes in MMRs, meaning a more sluggish pass-through. At the same time, a higher share of non-interest income or expenditures may make transmission stronger. Banks that are less dependent on interest-related income/ expenditures adjust interest rates more quickly in response to interbank rate changes. Acting in such a way banks try to capture market share in a competitive environment.

Administrative expenses and staff costs to gross income indicate the dependence of expenditures on MMR. Large administrative costs can reflect obstructive regulatory and legal frameworks, undeveloped financial infrastructure, and information asymmetry. To cover these excessive costs and preserve their margins, credit rates would be more responsive to increases in MMR. At the same time, if banks set their interest rates by adding a margin over their costs, then the proportion of costs that are not sensitive to interbank rate fluctuations would reduce the pass-through. A higher share of operating costs in total costs should imply a weaker transmission.

### 4. DATA

In this study, we interpret the strength of transmission as a response of banks' deposit or credit rates to interbank interest rates. Deposit and credit rates in the estimates are weighted average interest rates on term deposits of HHs in the national currency and weighted average interest rates on credits to NFCs in the national currency respectively. As an interbank rate, we use UONIA, which is the Ukrainian weighted average overnight interbank rate.

The strength of IR pass-through is estimated on monthly data. The samples used are different depending on the goals of the estimates. In ARDLs with constant parameters, we use a sample of 2015 m1 – 2023 m12, which covers the period of the inflation targeting<sup>5</sup> regime in Ukraine, meaning that the interbank rate significantly affected banking rates at that time. The estimates of the TVP ARDLs for the whole economy span a much longer period, which starts at 2006 m1. We use as long a sample as possible to give the Bayesian algorithm a reasonably large training dataset. For panel estimates, where we use bank-level data, the time sample is 2018 m1 – 2023 m12. The start point in 2018 is explained by the availability of reliable monthly data from the balance sheets of Ukrainian banks.

A balanced panel database was formed for the regressions that we use to analyze the transmission determinants. To avoid gaps in the data, we applied a selection procedure to the banks. First, the banks must be functioning during the sample of 2015–2023. Second, less than 30% of gaps in data were allowed for the time series of the banks' interest rates. Some of the banks did not perform credit or deposit operations in certain months. To keep the panel balanced, we filled these gaps by applying a linear spline interpolation.<sup>6</sup> After the selection procedure, we were left with 25 banks for the deposit rate transmission panel and 55 for credit rate transmission.<sup>7</sup>

# **5. RESULTS**

# **5.1. Linear and Asymmetric Estimates of the Pass-Through: Banking System Level**

Linear estimates of long-run pass-through for the whole banking system are presented in Figure 1. Deposit and credit rates are weighted averages of all banks that conducted respective deposit/credit operations in separate months. Figure 1 demonstrates impulse response functions to a 1% UONIA shock, which we derive from the linear ARDL model. The meaning of long-run transmission is the point at which the impulse response function converges. Our estimates show that long-run pass-through is higher for deposit rates, but that credit rates adjust faster.



Figure 1. Linear Estimates of Long-Run Transmission, 1% UONIA shock, %, month

<sup>&</sup>lt;sup>5</sup> Strictly speaking, after the start of the full-scale russian invasion (2022 m2) monetary policy in Ukraine switched to a mixed regime. Under this regime, in different periods exchange rate fixation and controlled exchange rate floating were primary goals. In this period, the NBU's interest rate was mainly used to maintain the attractiveness of domestic currency assets compared to the assets denominated in foreign currencies.

<sup>&</sup>lt;sup>6</sup> Definitions and basic statistics of variables can be found in Tables 4-5, Appendix A. Macroeconomic variables are the same for each bank. For this reason, the number of observations equals the number of months in the sample. Since the majority of variables coincide for both panels (transmission to credit rates and transmission to deposit rates), we report statistics from the credit rates transmission panel, which contains more banks.

<sup>&</sup>lt;sup>7</sup> 25 selected banks covered 96% of deposits in 2023, 55 banks – 99% of credits in 2023.

Asymmetric estimates (Figure 2) show us the difference in transmission depending on whether the UONIA changes are positive or negative. Asymmetry is more significant for deposit rates. The banks are usually more willing to cut deposit rates than to raise them (in part due to their wish to reduce interest costs). Therefore, the transmission strength of an increase in the UONIA is smaller than that of a decrease.

The asymmetry of transmission to credit rates is also downward, but not as significant as for deposit rates. Nevertheless, we would expect higher transmission for positive changes in interbank rates, due to the profit maximization motives of banks. Downward asymmetry can be caused by competition factors and the desire of a bank to occupy a larger share of the market. Another possible explanation is the size of a debtor. Big firms can have more favorable credit conditions compared to medium- and smallsize companies. Having high market power, big firms might be more aggressive in negotiating a cut in credit rates when market rates go down. To illustrate this possibility, we estimated the IRFs of credit rates for firms of different sizes (Figure 3). Our results show that downward asymmetry in credit rates pass-through mainly comes from the rates for large enterprises.

# 5.2. Bank-Level Estimates of the Pass-Through

Below, we present the outcomes of the estimated longrun market rate transmission (sample 2015m1–2023m12) at the bank level. Initially, our focus is on the long-run pass-through estimations for household deposit rates, obtained through both ARDL models and an asymmetric ARDL model.

For each bank, we ran the ARDL model with an optimal lag-structure based on the Schwarz criterion. Tables 6–7 report descriptions and testing statistics for the two sets of ARDLs used for the estimation of transmission to deposit and credit rates. In general, the ARDLs obtained indicate the existence of a long-run relationship between IR and respective rates and have adequate properties in terms of autocorrelation or heteroscedasticity of residuals<sup>8</sup>.

Figure 4 shows the histogram of the long-run passthrough estimations for the 25 commercial banks chosen in the previous section. In general, the pass-through for the majority of the banks falls within the range of 0.5 to 0.7, with a smaller portion exhibiting higher values of pass-through.

As can be seen in Figure 5, the long-run pass-through to deposit rates is higher for state banks and lower for the banks with foreign private capital. This is due to the influence of various factors. For example, state banks, being under public ownership or control, might have strategic motivations to quickly adjust their deposit rates in response to changes in market rates. On the other hand, banks with foreign private capital may have different risk appetites or operational strategies, leading to a comparatively lower pass-through.







Figure 3. Asymmetric Long-Run Transmission for (A) Large, (B) Medium, and (C) Small Firms, 1% UONIA shock, %, month

<sup>&</sup>lt;sup>8</sup> A long-run relationship was identified for 20 banks out of 25 for deposit rates, and 49 out of 55 for credit rates. Autocorrelation in residuals was rejected respectively for 19 and for 42 banks. Homoscedasticity was confirmed for 18 and 24 banks respectively.







Figure 5. Long-Run Pass-Through to Deposit Rates

Our estimates let us plot relationships between transmission strength and banks' characteristics via scatterplots. Figure 6 plots banks' pass-through and their share of the total amount of deposits. The latter is interpreted as a proxy for market power. Figure 6 indicates that for almost all foreign private banks, there is an increase

in transmission with a rise in deposit share. Notably, the state bank that holds nearly half of the deposits exhibits the highest pass-through. However, the relationship between the share of deposits and transmission for private domestic banks appears to be ambiguous, suggesting the presence of other significant factors that influence transmission.



Figure 6. Long-Run Pass-Through to Deposit Rates vs Deposit Shares

Figures 7 and 8 (Appendix B) show the results of the estimation of the NARDL model with asymmetry. For the majority of banks, the response to a decrease in the MMR is more pronounced, prompting a swift adjustment of deposit rates downward. Conversely, during rate hikes, there may be a tendency for a more gradual adjustment, as banks seek to avoid increased expenditures. A test of the statistical significance of the asymmetry indicates that none of the instances of positive asymmetry are statistically significant. Furthermore, the test did not identify significant asymmetry in the estimates for banks with high market power.

In Figure 9 (Appendix B), the histogram illustrates long-run pass-through estimations for the 55 commercial banks. The median pass-through to NFC loans surpasses that of deposits. Notably, for NFC loans, a more varied distribution is apparent, featuring cases with both low and high transmission at the extremes.

In contrast to deposit rates, private banks with foreign capital generally exhibit higher transmission for NFC loans than those with domestic capital (Figure 10 in Appendix B). One of the possible explanations for this may be that private banks with foreign capital might have diverse and potentially more extensive funding sources, impacting their flexibility in adjusting lending rates. This could contribute to a more pronounced transmission effect.

The impact of market power remains uncertain. It's worth noting that among all banks (excluding the largest one), those with a loan share above 0.5% exhibit long-run pass-through exceeding 0.65. Figure 11 illustrates that banks with exceptionally high or low pass-through rates typically have a rather small share of loans.

Much like the observed pattern in deposit rates, asymmetry is evident in many banks when it comes to NFC loan rates, with a larger share of banks having a more significant response when the MMR decreases (see Figure 12 in Appendix B). This may be explained by the fact that when the MMR decreases, banks may compete more vigorously for borrowers. As a result, they may be more inclined to adjust loan rates downward promptly to attract and retain customers.

Concerning the statistical significance of the asymmetry, we observe more cases for loan rates: the asymmetry is significant for almost half of all banks (see Figure 13 in Appendix B) and remains notable even among banks with high market power.

To get a general picture of the differences in transmission for banks with different forms of ownership, we ran panel ARDLs for three groups of banks: state, foreign, and domestic private banks. The IRFs presented in Figure 14 (Appendix B) demonstrate that there is no statistically significant difference in transmission to deposit rates for banks with different forms of ownership. However, in the long-run pass-through transmission in state banks is somewhat higher. Transmission to credit rates is much faster and stronger for foreign banks.

# 5.2. Time-Varying Pass-Through: Aggregated and Bank-Level Estimates

When estimating TVP transmissions we used ARDL specifications for each bank presented in Tables 6-7. Doing this, we assume that only the coefficients of the regression vary in time, while the specification optimized on the whole sample is constant. We consider this assumption as a good trade-off between the simplicity of estimations and the real frequency of changes in the decision rules of the banks. For each TVP-ARDL we estimated time-varying IRFs, which allows us to explore pass-through effects across different horizons. To keep our analysis parsimonious, we use the responses of the banks' rates after 3, 6, 9, and 12 months, and over the long run.

Figure 15 (Appendix B) shows estimations of timevariant long-run pass-through for household deposit rates for 25 banks. While the magnitude of pass-through may vary, the majority of banks exhibit a consistent dynamic pattern: a decline in long-run pass-through since the end of 2019. Nevertheless, from the second half of 2022 onward, there is an observable increase in pass-through – possibly influenced by the measures taken by the NBU to enhance the transmission from the MMR to deposit rates.

Apart from calculating the median pass-through, we also computed the pass-through weighted by the share of deposits. Throughout the entire period, the weighted pass-through consistently appeared lower than the median, suggesting that the largest market participants likely had below-average values of pass-through. This result contrasts with the findings obtained through ordinary ARDL, justifying further investigation. It is noteworthy that asymmetry ARDL yielded results more closely aligned with those from the TVP model.

The magnitude of TVP pass-through for NFC loan rates was found to be higher than that for deposit rates (see Figure 16 in Appendix B). To better comprehend the factors influencing transmission, the obtained TVP pass-through will be subsequently utilized in regression analyses to identify the determinants contributing to the transmission dynamics.

In Appendix B we present more results from TVP estimates. The medians of pass-through strength for different horizons (Figure 17) are positively correlated. This reflects the fact that the IRFs we obtained at the bank level are mostly smooth and monotonic. Kernel densities (Figure 18) indicate that in the short-run both transmissions are close to bimodal distribution, meaning that in our sample there is a sizable group of bank-month estimates that are significantly different from the general median. Over the 12-month horizon bimodality disappears and the median values of the transmissions for deposit and credit rates are close to each other.

The dataset contains several banks that have changed their ownership status between 2015 and 2023.<sup>10</sup> These banks might be interesting objects for separate studies on possible changes in transmission after their ownership changes. However, this question remains out of the scope of

<sup>&</sup>lt;sup>10</sup> Three banks in the sample for transmission to deposit rates: one changed its status from foreign private to domestic private, one from foreign private to state, and one from private domestic to state. Two banks in the sample for transmission to credit rates: one changed its status from foreign private to state, and one from private domestic to state.

our paper. Figure 19 shows time-varying transmissions across different horizons for the banks where ownership changed. The visual analysis does not indicate significant shifts in transmissions after the ownership transition, compared to the average dynamics of the market.

# **5.3.** Panel Estimates of the Pass-Through Determinants

The results of all of the panel regressions are reported in Appendix A. In Tables 8 - 9 we present regressions for long-run pass-through, which demonstrate our results in general. The specifications we report are regressions containing only bank-level variables, regressions containing only macroeconomic data, and joint estimates. To control for possible autocorrelation and heteroscedasticity in errors, we apply three estimators to each type of specification. In regressions with one-way clusterization, standard errors of coefficients and statistics are robust to both arbitrary heteroscedasticity and arbitrary correlation inside the panel units (banks). The option of two-way clusterization provides us with standard errors that are robust to arbitrary withinpanel autocorrelation and arbitrary contemporaneous crosspanel correlation. The third option we use to get consistent standard errors is the application of the Newey-West variance estimator, which gives statistics that are robust to both arbitrary heteroscedasticity and arbitrary autocorrelation. First of all, we pay attention to statistically significant variables in regressions (7) - (9) since they contain the full list of variables. Considering that cluster-robust algorithms can give inconsistent results when the number of clusters is small, we prioritize the results of the Newey-West variance estimator.

We also report regressions for different transmission horizons (Tables 10–11) to study whether some determinants can be significant in the short run. To keep the tables to a reasonable size only regressions with both bank-level and macroeconomic variables are included.

For a deeper understanding of potential cluster heterogeneity, we estimated baseline panel regressions for groups of banks. The banks were divided on the basis of ownership, into state-owned and privately owned. We also applied alternative grouping procedures. Hierarchical clustering using the Lance-Williams algorithm was conducted to cluster banks based on the correlations between the TVP monetary transmission of different banks, for both deposits and credits. Due to sample limitations, only the group with the largest cluster (K1) was considered. The main results for group estimations will be discussed below, with further details available in Figures 20–21.

#### **Transmission to Deposit Rates**

Transmission to deposit rates is positively affected by the share of the bank in the deposit market (*hhd\_shr*). This result corresponds to the coefficient of market concentration (*hhi\_dep*), which is positive. Bigger banks demonstrate higher transmission strength, and a more concentrated market means higher deposit rate transmission. This somewhat counterintuitive result might be explained by the desire of big banks to occupy more market share, which forces them to demonstrate higher transmission. Another possible reason for such a result is the existence of two big players on the market with high transmission estimates (Figure 6). These outliers can bias the general picture significantly. The coefficient of net interest income to assets ratio (*intrev\_a*) has a negative sign and is statistically significant for short horizons. We relate this to a positive correlation between interest expenses and the strength of transmission. More responsive banks pay higher interest rates (at least in periods of interbank rate increase), thus decreasing net interest income.

Provisions to assets ratio (*res\_a*) indicate that the banks with more risky assets actively change deposit rates. That may happen because of the desire to attract new funds from households.

Costs paid for funding from households ( $fu\_cst\_hh$ ) are also positively correlated with transmission strength, reflecting the accounting effects of changes in deposit rates. When transmission is high, banks actively increase\ decrease deposit rates in response to respective changes in IR and pay higher\lower costs for acquired funds. Funding costs for resources acquired from the NFCs ( $fu\_cst\_nfc$ ) are insignificant in the final specification. It means that the policies of the banks on setting rates for HHs and NFCs are independent of each other.

Ratios of administrative and staff costs (*nadmexp\_exp*, *persexp\_exp*) have positive coefficients, although they are mostly insignificant. These expenses are independent of interest rate fluctuations, meaning that when their shares increase a bank can be more responsive to interbank rate changes without seeing considerable effects on overall expenditures.

The positive effect of the share of interest income (intrev\_rev) is an indication of the compensatory behavior of the banks. Having higher transmission to deposit rates (and higher interest expenses respectively), banks try to adjust credit rates more actively to cover the additional costs.

High liquidity negatively affects transmission strength, confirming the assumption that it reduces incentives for banks to attract new funds. On the bank level, we define liquidity as cash and cash equivalents to assets ratio ( $liq_a$ ), while on the macroeconomic level liquidity is the ratio of CDs on banks' balance sheets to liabilities (liq). These two measures of liquidity can be used interchangeably in regressions. When we remove the macroeconomic measure of liquidity from regression, bank-level liquidity negatively affects transmission as well (Table 12, regressions 1–3).

The growth of economic uncertainty (*fsi*) decreases transmission, making banks more cautious in rate adjustments. An increase in deposit rates can lead to extra costs for banks, which are undesirable in times of macroeconomic volatility. A decrease in rates can lead to a fast outflow of deposits.

Deposits supply (*rdep\_gr*) decreases transmission, demonstrating that banks can get more funding for less cost when deposit inflows are high. CPI has a negative sign in all regressions, contradicting the hypothesis that banks have to maintain real deposit rates at some level via indexation of nominal rates. On the one hand, the negative sign of CPI can be explained by economic uncertainty, which is not fully captured by FSI. In this case, CPI demonstrates uncertainty effects on transmission, when high inflation happens in times of turbulence. On the other hand, the effect of CPI can be nonlinear, depending on the degree of inflation. To show this, we have added the square of CPI in regressions (Table 12, regressions 4–6). Nonlinear modification of regression

demonstrates that the impact of CPI becomes positive when inflation exceeds 23.3% (sample mean for 2018m1-2023m12 equals 10.8%, maximum -26.6%).

The reserve requirement ratio for households' deposits on demand and current accounts (*norm\_res*), together with the share of long-term deposit certificates (*dc\_100d\_shr*), have a positive effect on transmission. These variables are interpreted as instruments used by the NBU to enforce the pass-through. According to the estimates, the impact of the reserve requirement ratio and the attractiveness of long-run NBU deposit certificates are in line with our assumptions.

The results for different groups of banks (Figure 20) show that most factors influence transmission in the same direction, but that the strength of the influence may differ. For example, for state-owned banks, the reserve requirement ratio seems to have a greater influence than the share of long-term deposit certificates, while the opposite is true for private banks. Additionally, inflation has a positive sign in the state-owned group, which aligns with theoretical reasoning.

### **Transmission to Credit Rates**

In the regressions, we have three variables that are proxies for competition in the credit market. Transmission to credit rates becomes stronger with the growth of the market share (*nfcl\_shr*) of a bank. The credit market concentration indicator (*hhi\_cred*) also has a positive coefficient, meaning that lower competition leads to higher transmission. Net interest income to assets ratio (*intrev\_a*) positively affects transmission. The banks generating high net interest rate margins adjust their credit rates more actively (and may be less sensitive in deposit rates (Table 8)) than those with low margins. These effects demonstrate that the market power of a bank increases its ability to adjust credit rates.

A high share of NPLs  $(npl_agr)$  restricts the reaction of banks to IR fluctuations. The negative effect is explained by liquidity requirements having to be fulfilled, and the need to heal the credit portfolio rather than expanding it. The proportion of NPLs is an indicator of already realized credit risks, which have to be covered by liquidity rationing as resultantly less aggressive policy on the credit market. At the same time, the provisions of the banks approximate the magnitudes of potential risks. The growth of the provisions to assets ratio (*res\_a*) indicates an increase in credit portfolio riskiness. A positive coefficient in the regressions indicates that the banks are covering high risks with higher rates and more aggressive reactions to IR changes, increasing pass-through.

A high capital ratio (*vk\_liab*) increases the strength of transmission, especially in the short run. Playing the role of a safety buffer, the higher the level of a bank's capital, the more aggressive it can be on the credit market, adjusting interest rates by more. The high share of term deposits (*termdep\_liab*) in liabilities provides banks with stable funding costs, making them less vulnerable to changes in market rates and thus leading to weaker transmission to credit rates.

A positive sign for interest expense share (*int\_exp*) signals a connection between the credit and deposit rates policies of the banks. As in regressions for deposit rate transmission, banks demonstrating higher interest expenses try to compensate for these costs by actively adjusting credit rates.

High liquidity in the banking system (*liq*) negatively affects transmission. Extra liquidity reflects the risk perception of the banks or low opportunities for new credits. When risks in an economy are considered extremely high, banks prefer more safe and liquid assets and react weakly to IR movements.

Macroeconomic uncertainty (*fsi*) makes transmission stronger. Banks try to account for increased financial risks by actively adjusting credit rates.

Inflation has an insignificant effect on transmission. Nevertheless, the coefficient is negative, which might be explained (as in the case of transmission to deposit rates) by nonlinearities. The inclusion of squared inflation (Table 13) gives a threshold of about 20%, after which banks become more sensitive to inflation, adjusting credit rates.

The variable of credit demand (*rcred\_gr*) has a positive sign. In times of high demand, the banks have the opportunity to increase credit rates by more. When demand for credits goes down, banks try to attract clients by lowering rates more aggressively.

Variables testing for the existence of the crowding-out effect (*ovdp\_a*, *dc\_100d\_shr*) have positive coefficients, although only the coefficient of the long-run deposit certificates share is statistically significant in all regressions. This result confirms the crowding-out effect of the relatively long securities allocated by the government and the NBU.

The difference between estimation results for various groups is more pronounced for credit rates compared to deposit rates. For example, for state banks, factors such as lower competition and high capital ratio are more important, whereas excess liquidity does not play a significant role. Conversely, the factor of the crowding-out effect can be observed for private banks and banks from the K1 cluster.

### **Discussions and Extensions**

Our study relies on estimates of transmission strength obtained from ARDL models. This approach has the advantage of simplicity, but it also assumes the exogeneity of the MMR with respect to the banks' rates. This assumption could be too strong, as there is evidence of policy rate endogeneity (Saborowski and Weber, 2013). The estimation of the effects with a VAR model may partially tackle the endogeneity issue, yet bring its own set of challenges. The analysis of transmission determinants also could be conducted in a VAR setup by running PVAR models. We recognize the possibility of another approach to the study of transmission in Ukraine.

In some cases, the results we obtained in the analysis of factors affecting transmission are not coherent with the conventional outcomes of other studies. For instance, the results from the regressions for both transmissions suggest that banks' policies regarding the adjustment of credit and deposit rates are positively correlated. Increasing interest expenses in relative terms results from high transmission to deposit rates. To cover high expenses, the banks add a margin over deposit rates, which leads to higher transmission to credit rates. Tables 14-15 explicitly include credit rate transmissions into regressions for deposit rate transmission and vice versa. The results show that deposit rate transmission is positively correlated with credit rate transmission over different horizons (3, 6, 9, 12 m) confirming that banks with increasing transmission to deposit rates also increase transmission to credit rates.

Transmission to deposit rates is positively affected by market concentration. This result could be driven by two outliers, which are state-owned banks occupying large shares of the market and demonstrating relatively high transmission (Figure 6). To test for this, we excluded these two banks from our estimates (Table 16). The results show that market concentration now negatively affects transmission, as seen in the majority of empirical studies. Considering this, we conclude that the two large banks distort the overall picture. On average, the majority of the banks decrease transmission to deposit rates when the market becomes more concentrated. This finding turns us to the question of the strategies that are used by different banks in setting interest rates. Some banks can increase deposit rates aggressively, demonstrating higher transmission and getting higher shares of the market. If such behavior is true, market concentration will be positively correlated with transmission strength. On the other hand, a negative coefficient of market concentration in regressions can indicate a situation in which big banks actively adjust deposit rates down in response to IR decreases (transmission goes up) and lose some share of the market as their deposits become less attractive. These examples show possible nonlinearity in the relationship between market concentration and transmission strength, depending on the sign of the IR change.

Tables 17–18 present regressions with tests for nonlinear effects of market concentration on transmission to deposit and credit rates depending on the increase/ decrease of the IR. We tested variables of market shares (*hhd\_shr*, *nfcl\_shr*) and market concentration (*hhi\_dep*, *hhi\_cred*) with a dummy of an IR decrease. The estimates indicate that nonlinearity is more statistically significant for transmission to deposit rates. With market concentration growth, deposit rates become more responsive to an IR decrease. The same effect, though non-significant, is present for the market shares of banks. For credit rates, an increase in market concentration or market share makes banks less responsive to an IR decrease. However, these nonlinearities are weak.

The negative impact of inflation on transmission in linear specification requires deeper analysis. Lower transmission to deposit rates in times of high inflation might be explained by the desire of banks to minimize real interest expenses. But this explanation fails to account for credit rates, as banks must try to maximize real interest incomes. For a better description of transmission, we introduced threshold effects of inflation and expanded our sample to 2015m1-2023m12. The years 2015-2016 are a period of high economic turbulence, with high inflation. This data extension gives us a longer sample for a better understanding of inflationary effects on transmission and higher volatility of inflation for better identification of nonlinear effects. Tables 19–20 contain regression specifications with macroeconomic variables only, as banklevel data is absent for such a long sample. To test for the threshold effects, we include a squared term of CPI. In extended sample results, inflation changes its coefficient to positive in regressions for transmission to deposit rates, while retaining a negative coefficient in regressions for credit rate transmission. The main point is that for the 2015-2023 sample, the effects of inflation lose statistical significance. However, estimates with threshold effects of inflation have high statistical significance and give a threshold of about 16% for both transmissions. A negative

impact on transmission under the threshold means that banks try to minimize real interest expenses by adjusting deposit rates and try to retain clients by setting lower credit rates. Over the threshold, banks start to reset deposit rates more actively to retain clients and adjust credit rates by more to maintain real interest income.

## **6. CONCLUSIONS**

Interest rate pass-through in Ukraine shows different patterns for deposit and credit rates. On average, long-run transmission is higher for deposit rates, but banks adjust credit rates more quickly. Transmission is highly nonlinear when it comes to an increase/decrease in the interbank rate. The responses of both deposit and credit rates are downward asymmetrical. For deposit rates, this is explained by the desire of banks to minimize funding costs. Credit rates can be more responsive to an IR decrease because of competition between the banks and the negotiating power of big borrowers.

Whatever their form of ownership, banks on average demonstrate close magnitudes of pass-through – except in transmission to credit rates of foreign banks, which is significantly higher compared to other banks.

The strengths of transmissions to deposit and credit rates are positively correlated. Banks that demonstrate high transmission to deposit rates are also more responsive to IR fluctuations in the credit market.

Time-varying estimates of the long-run pass-through on the bank level allow us to investigate the effects of balance sheet and macroeconomic variables on transmission. Our results show that market concentration increases the responsiveness of deposit and credit rates. For deposit rates, this result is driven by two outliers, which dominate the deposit market and demonstrate high transmission to deposit rates. The exclusion of these outliers yields a negative relationship between market concentration and transmission to deposit rates. We interpret this as evidence that big players try to occupy more market shares by actively decreasing/increasing credit/deposit rates or trying to get monopolistic rent increasing/decreasing rates for credits/ deposits when reacting to the respective movements of the interbank rate. However, the rest of the banks demonstrate lower transmission in a concentrated market.

The relationship between market concentration and transmission is highly nonlinear depending on the increase/ decrease of the IR. When market concentration goes up, deposit rates become more responsive to an IR decrease, while credit rates demonstrate lower transmission effects.

Increasing the share of NPLs decreases the passthrough to credit rates, preventing banks from accumulating new credits and making them pay more attention to financial sustainability parameters. The riskiness of assets increases transmission strength to both deposit and credit rates. For credit rates, this indicates a more aggressive policy of a bank on the credit market, which, in turn, requires more flexible adjustment of deposit rates to attract additional funding resources. Funding costs are associated with higher transmission to deposit rates, meaning that on average banks need to pay additional expenses for flexibility in the market of deposits. In the short run, the pass-through to credit rates becomes stronger with an increase in the equity capital to assets ratio. Playing the role of a safety buffer, a high capital ratio allows a bank to take on more credit risks. Banks that increase the share of term deposits in their liabilities are less vulnerable to changes in market rates, which leads to weaker transmission to credit rates.

At the macro-level high liquidity negatively affects transmission to both rates. For the deposit market, extra liquidity reduces incentives for banks to attract new funds, while on the credit market it reflects the unwillingness of banks to lend because of high risks. Macroeconomic uncertainty decreases transmission to deposit rates and positively affects transmission to credit rates. In times of high economic volatility, banks try to minimize their costs, and as a result, deposit rates are weakly adjusted. On the other hand, uncertainty is associated with higher credit risks, which banks try to cover with higher rates. Inflation has a statistically significant negative effect on transmission to deposit rates. Moderate inflation allows banks to pay less for deposits in real terms, so rates are rigid. However, we found nonlinear effects: banks start to adjust deposit rates

# REFERENCES

Apergis, N., Cooray, A. (2015). Asymmetric interest rate pass-through in the U.S., the U.K. and Australia: New evidence from selected individual banks. Journal of Macroeconomics, 45(C), 155–172. https://doi.org/10.1016/j.jmacro.2015.04.010

Aristei, D., Gallo, M. (2014). Interest rate pass-through in the euro area during the financial crisis: A multivariate regimeswitching approach. Journal of Policy Modeling, 36(2), 273–295. https://doi.org/10.1016/j.jpolmod.2013.12.002

De Bondt, G. (2005). Interest rate pass-through: Empirical results for the euro area. German Economic Review, 6(1), 37–78. https://doi.org/10.1111/j.1465-6485.2005.00121.x

Deutsche Bundesbank (2019). Interest rate pass-through in the low interest rate environment. Monthly Report, April 2019, pp. 33–35.

Dieppe, A., van Roye, B., Legrand, R. (2016). The BEAR toolbox. Working Paper Series 1934. Frankfurt: European Central Bank. Retrieved from https://www.ecb.europa.eu/pub/pdf/scpwps/ecbwp1934.en.pdf

Fičura, M., Witzany, J. (2023). Determinants of nonmaturing deposit pass-through rates in Eurozone countries. Politická ekonomie, 3, 291–318. https://doi.org/10.18267/ j.polek.1388

Filatov, V. (2021). A new financial stress index for Ukraine. Visnyk of the National Bank of Ukraine, 251, 37–54. https://doi.org/10.26531/vnbu2021.251.03

Gambacorta, L., lannotti, S. (2007). Are there asymmetries in the response of bank interest rates to monetary shocks? Applied Economics, 39(19), 2503–2517. https://doi.org/10.1080/00036840600707241

Gigineishvili, N. (2011). Determinants of interest rate pass-through: Do macroeconomic conditions and financial market structure matter? IMF Working Papers, 11/176. Washington: International Monetary Fund. Retrieved from https://www.imf.org/en/Publications/WP/Issues/2016/ 12/31/Determinants-of-Interest-Rate-Pass-Through-Do-Macroeconomic-Conditions-and-Financial-Market-25096 more actively when inflation exceeds 16%. This threshold also works for transmission to credit rates. The growth of deposit supply decreases pass-through, allowing banks to acquire more deposits at lower rates. High demand for credits increases transmission to credit rates.

The results of panel regressions for transmission to deposit rates confirm that the measures taken by the NBU to encourage pass-through had a significant positive effect. The increased reserve requirement ratio for households' deposits on demand and current accounts, together with the ability to allocate liquidity to long-term deposit certificates, made the banks adjust their term deposit rates.

We also found that having government bonds in assets and having a higher share of long-term deposit certificates make transmission to credit rates stronger. We interpret this result as evidence of the crowding-out effect, as borrowers have to compete with government and monetary authorities for credit resources.

Gregor, J., Melecky, M. (2018). The pass-through of monetary policy rate to lending rates: The role of macro-financial factors. Economic Modelling, 73(C), 71–88. https://doi.org/10.1016/j.econmod.2018.03.003

Hlazunov, A., Dadashova, P., Lukianenko, I. (2023). Interest rate pass-through in Ukraine: evidence from the bank ownership. Financial and Credit Activity Problems of Theory and Practice, 5(52), 8–24. https://doi.org/10.55643/ fcaptp.5.52.2023.4135

Kourelis, A., Cottarelli, K. (1994). Financial structure, bank lending rates, and the transmission mechanism of monetary policy. IMF Working Papers, 1994/039. Washington: International Monetary Fund. https://doi.org/10.5089/9781451845761.001

Leroy, A., Lucotte, Y. (2016). Structural and cyclical determinants of bank interest-rate pass-through in the Eurozone. Comparative Economic Studies, 58(2), 196–225. https://doi.org/10.1057/ces.2016.6

Leuvensteijn, M., Kok, C., Sørensen, J., Bikker, A., van Rixtel, A. (2008). Impact of bank competition on the interest rate passthrough in the euro area. ECB Working Paper, 885. Frankfurt: European Central Bank. Retrieved from https://www.ecb.europa.eu/pub/pdf/scpwps/ecbwp885.pdf

Ma, J., Wang, H. L. (2014). Theoretical models for policy rate transmission mechanism. People's Bank of China Working Paper, 2014/1.

Månsson, K., Shukur, G., Sjölander, P. (2013). Asymmetric quantile analysis of the Swedish mortgage price discovery process. Applied Economics, 45(21), 3088–3101. https://doi.org/10.1080/00036846.2012.681030

Mishra, P., Montiel, P. (2013). How effective is monetary transmission in low-income countries? A survey of the empirical evidence. Economic Systems, 37(2), 187–216. https://doi.org/10.1016/j.ecosys.2012.12.001

Reinhart, C., Rogoff, K. (2004). The modern history of exchange rate arrangements: A reinterpretation.

The Quarterly Journal of Economics, 119, 1, 1–48. https://doi.org/10.1162/003355304772839515

Saborowski, C., Weber, M. S. (2013). Assessing the determinants of interest rate transmission through conditional impulse response functions, IMF Working Papers, 13/23. Washington: International Monetary Fund. Retrieved from https://www.imf.org/external/pubs/ft/wp/2013/wp1323.pdf

Sander, H., Kleimeier, S. (2002). Asymmetric adjustment of commercial bank interest rates in the euro area: an empirical investigation into interest rate pass-through. Credit and Capital Markets, 35, 161–192. https://doi.org/10.3790/ ccm.35.2.161

Sander, H., Kleimeier, S. (2004). Convergence in eurozone retail banking? What interest rate pass-through tells us about monetary policy transmission, competition and integration. Journal of International Money and Finance, 23(3), 461–492. https://doi.org/10.1016/j.jimonfin.2004.02.001

Shin, Y., Yu, B., Greenwood-Nimmo, M. (2014). Modelling asymmetric cointegration and dynamic multipliers in a

nonlinear ARDL framework In: Sickles, R., Horrace, W. (eds) Festschrift in Honor of Peter Schmidt, 281–314. https://doi.org/10.1007/978-1-4899-8008-3\_9

Stanisławska, E. (2015). Interest rate pass-through in Poland: Evidence from individual bank data. Eastern European Economics, 53(1), 3–24. https://doi.org/10.1080/ 00128775.2015.1033362

Sznajderska, A. (2012). On the empirical evidence of asymmetry effects in the interest rate pass-through in Poland. NBP Working Papers, 114. Warsaw: Narodowy Bank Polski. Retrieved from https://static.nbp.pl/publikacje/materialy-istudia/114\_en.pdf

Terrier, G., Valdes, R., Tovar Mora, C., Chan-Lau. J., Valdovinos. C. F., Garcia-Escribano, M., Medeiros, C., Tang, M-K., Martin, M., Walker, C. (2011). Policy instruments to lean against the wind in Latin America. IMF Working Papers, 2011/159. Washington: International Monetary Fund. Retrieved from https://www.imf.org/-/media/Websites/ IMF/imported-full-text-pdf/external/pubs/ft/wp/2011/\_ wp11159.ashx

# **APPENDICES**

# **APPENDIX A. TABLES**

Table 2. Datasets and Methods used in Interest Rate Pass-Through Studies

Sources	Sample	Transmission	Method
Saborowski and Weber (2013)	142 countries, unbalanced panel data over the sample period 2000m1:2011m12.	Central bank policy rates to retail lending rates.	Panel interaction VAR
Leroy and Lucotte (2016)	Sample of 11 euro area countries over the period 2003m1:2011m12.	MMR to harmonized bank lending rates to households and firms.	Single equation ECM, panel interaction ECM, panel interaction VAR
Gigineishvili (2011)	70 countries including low- income, emerging, and developed economies. Monthly data, 2005m12:2010m3.	From three-month treasury bill rates to one- to two-year corporate loans in domestic currency.	ARDL, Cross-sectional regression
Mojon (2000)	6 largest countries in the euro area (Belgium, Germany, Spain, France, Italy and the Netherlands). Annual averages over the periods 1979-82, 1982- 88, 1988-92 and 1992-98.	From MMR to 25 credit rates and 17 deposit rates.	ECM, Panel regression
Gregor and Melecký (2018)	Czech Republic, 2004m1:2017m11.	From CNB policy rate to four lending rates: the consumer loan rate, mortgage loan rate, small corporate loan rate, and large corporate loan rate.	ARDL with interaction terms
Cottarelli and Kourelis (1994)	31 industrial and developing countries. Monthly data from the 1980s to 1993.	From MMR to the lending rate	Distributed lag regression, Cross-sectional regressions
Sørensen and Werner (2006)	10 countries, 1999m1:2004m6	Speed of transmission from market interest rates to 4 series of lending rates (loans to households for consumption, loans to households for house purchase, short-term loans to non-financial corporations, and long-term loans to non- financial corporations) and 3 series of deposit rates (current account deposits, time deposits, and savings deposits.	ECM by DSUR
Leuvensteijn et al (2008)	8 euro area countries. 1992– 2004	From three-month MMR or government bond yields to bank loans interest rates	Panel ECM
Stanisławska (2015)	Poland, bank-level data, 2005m1 – 2013m7	From MMR with maturity of 1 month and 3 months to deposits of households and firms as well as loans granted to households and firms	Panel ECM

Sources	Factors	Variable	Impact on transmission	Empirical findings
Saborowski and Weber (2013)	Regulatory quality	World Bank regulatory quality index.	A weak regulatory environment creates uncertainty in the financial system and leads to a deformalization of financial transactions and a higher cost of financial intermediation. As a result, bank rates become less sensitive to changes in the policy rate.	Significant and sizable negative influence on pass- through.
Saborowski and Weber (2013)	Financial dollarization	Share of foreign currency Ioans in total Ioans	The cost of foreign currency funding is linked to external factors that are mostly outside the control of the central bank. To the extent that financial market participants can arbitrage between domestic and foreign currency instruments, the policy rate can thus only partially control market interest rates on domestic currency instruments.	Significant and sizable negative influence on pass- through.
Cottarelli and Kourelis (1994), Gigineishvili (2011), Saborowski and Weber (2013), Leroy and Lucotte (2016)	Financial development	Private sector credit and deposits as a share of GDP, Market capitalization to GDP ratio, Total assets and liabilities, Per capita GDP, The ratio between broad money and GDP (M2/ GDP), The ratio between broad and narrow money	More variety in investment opportunities leads to increased competition between financial products. Market interest rates – including on wholesale markets – are thus more responsive to policy rate changes because profit margins are constrained.	Significant and large influence on the effectiveness of interest rate transmission.
Sørensen and Werner (2006), Gigineishvili (2011), Saborowski and Weber (2013), Stanisławska (2015)	Liquidity ratio	Share of liquid assets in total assets and short- term liabilities.	Countries with low excess liquidity have higher pass-through. Excess liquidity reflects differences in risk perceptions or a more conservative business model. It is also related to the lack of investment opportunities. In excessively liquid markets, when all banks are structurally on the same side of the market, interbank trading in short-term funds dries up, and interest rates fail to reflect the true marginal cost of financial resources. Naturally, the retail pricing of loans becomes less responsive to MMR and the connection between the two weakens.	Low excess liquidity enforces pass-through.
Cottarelli and Kourelis (1994), Mojon (2000), Sørensen and Werner (2006), Leuvensteijn et al (2008), Saborowski and Weber (2013), Leroy and Lucotte (2016), regor and Melecký (2018)	Competition	Banking sector concentration (Herfindahl index computed over asset shares of individual commercial banks), The banking sector's average return on equity, The standard deviation of NPLs across banks, Lerner Index of market power, The index of deregulation measures taken by European countries between 1980 and 1995,	In countries with a low concentration of the banking sector, long-run pass- through is higher. When banks have substantial market power, policy rate changes may translate into movements in spreads rather than market rates.	Banking competition reinforces the transmission of monetary policy. Banks tend to price their loans more in accordance with the market in countries where competitive pressures are stronger. Stronger loan market competition also means larger bank spreads (implying lower bank interest rates) on current accounts and time deposits. This would suggest that the competitive pressure is heavier in the loan market

Table 3. Potential Determinants to Interest Rate Pass-Through Analyzed	in the Literature
--	-------------------

			5 ,	
Sources	Factors	Variable	Impact on transmission	Empirical findings
		Market share of the largest five banks and the number of bank branches per 100,000 inhabitants, Market power (RoE), Boone indicator		than in the deposit markets, so banks compensate for the reduction in their loan market income by lowering their deposit rates.
Sørensen and Werner (2006), Gigineishvili (2011), Saborowski and Weber (2013), Stanisławska (2015), Leroy and Lucotte (2016).	Asset quality	Share of non-performing in total loans, Aggregate measure of distances to default of listed banks by country, Gross of provisions.	Banks with weak balance sheets may react to an expansive monetary policy stance by shoring up liquidity rather than extending credit at lower rates. A change in the policy rate may thus have only a limited impact on market rates. Moreover, banks have to comply with regulatory requirements, implying that their capacity to expand lending depends, for instance, on their capital adequacy. However, the ratio of provisions to gross income is often negatively correlated with concentration ratios. Hence, this might imply that banks in a more competitive environment on average take on more problematic loans (leading to higher provisions), which might explain the positive effect on the pass-through.	Results are mixed. The majority of studies indicate that countries/banks with low NPLs have a long-term pass-through that is higher than that of countries/ banks with high NPLs.
Gigineishvili (2011), Saborowski and Weber (2013)	Exchange rate regime	Rigidity of exchange rate regimes from classification by Reinhart and Rogoff (2004)	The central bank's control over market rates is likely to be tighter when policy rates are set as part of a transparent and rules-based framework that is largely independent of other influences such as fiscal and exchange rate policy. A lack of exchange rate flexibility may signal that the policy rate is not set with the primary purpose of steering an intermediate target of monetary policy such as market interest rates or commercial bank reserves.	Countries with rather rigid exchange rate regimes have a long-run pass- through lower than those with flexible rates.
Cottarelli and Kourelis (1994), Mojon (2000), Gigineishvili (2011), Saborowski and Weber (2013), Leroy and Lucotte (2016)	Inflation	Inflation rate	Higher inflation ratios lead to more frequent price changes, thus making it more likely that policy rate changes impact market rates in a timely fashion.	Results are mixed. The majority of studies indicate that inflation increases pass-through.
Mojon (2000); Gigineishvili (2011); Leroy and Lucotte (2016)	Business cycle, Deposits supply, Loans demand	Industrial production index, The average real growth rate of credit and deposit volumes for retail markets, Average real growth rate of GDP or residential or non-residential investment, Gross national saving ratio, House prices.	An increase (decrease) of industrial production would encourage banks to modify the allocation of their portfolios towards riskier (low- risk) projects. However, since the perceived credit risk of borrowers is pro-cyclical, a reverse effect would also be possible. GDP and credit growth also can have a negative effect on the speed of adjustment. An increase in loan demand/ deposit supply allows banks to reduce the speed of interest rate adjustment, although the results are not unambiguous.	Decrease of pass-through during a crisis and in low- income economies. The volume of credit and real demand both tend to lower the pass-through to credit rates when interest rates are falling, while their impact is not significant when interest rates are rising. This is partially consistent with the ability of banks to preserve their interest rate margin on credit when they face stronger credit demand.

### Table 3 (continued). Potential Determinants to Interest Rate Pass-Through Analyzed in the Literature

Sources	Factors	Variable	Impact on transmission	Empirical findings
				When rates increase, higher saving ratios seem to allow banks to adjust deposit rates faster than when the rates decrease, while higher GDP growth has the opposite effect.
Leroy and Lucotte (2016)	Macro-financial stance	Stock price Index; Spread between swaps 6-month Euribor and 10- year government bond rates; Spread between specific government bond rate and the EONIA.	A higher stock price index increases the pass-through and opposite effects for the country risk premium because of the marginal cost of bank funding.	Positive effect of stock price index on pass- through. Country risk premium alters the pass-through.
Cottarelli and Kourelis (1994), Mojon (2000), Gigineishvili (2011), Leroy and Lucotte (2016)	Markets volatility	MMR volatility; Size of the "random component" in the MMR series.	MMR volatility is expected to have a negative effect on bank interest pass-through. Theory suggests that monetary transmission will be more important whether the bank is confident that the change is permanent, and not temporary.	The negative impact of the MMR volatility on pass- through.
Sørensen and Werner (2006), Gigineishvili (2011)	Bank capitalization	Capital to total assets and capital to risk- weighted assets.	Excess capital at banks may act as a buffer against market fluctuations and would hence be expected to show a negative relation with the speed and strength of adjustment.	Negative effect on the speed of pass-through
Gigineishvili (2011)	Profitability	Returns on equity and assets; Net interest margin.	High profitability in the banking sector, which is often a reflection of inadequate competition and market power, weakens pass-through. In an uncompetitive environment, banks can charge higher premiums and deviate from marginal cost pricing. As a result, lending rates become less elastic to changes in the costs of funds.	Negative effect on pass- through
Mojon (2000), Gigineishvili (2011)	Costs structure	Overhead costs to total assets	Large overheads could be structural in nature, related to obstructive regulatory and legal frameworks, undeveloped financial infrastructure, and information asymmetry. These may include excessive costs originating from burdensome registration requirements, prolonged court litigation, legal obstacles in seizing and liquidating collateral, difficulties in assessing the creditworthiness of borrowers and the value of collateral, and so forth. To cover such costs, banks would have to maintain higher interest margins. To preserve these margins, an increase in MMR is likely to induce stronger pass-through to lending rates.	The coefficient for overhead costs is positive and statistically significant, implying that overhead costs contribute to the strength of pass-through.
		Ratios of banks' staff costs to banks' gross income.	If banks set their interest rates by adding a margin over their costs, one can expect the pass-through to reflect the impact of changes in the MMR on the total costs of the bank. A priori, a higher share of operating costs in total costs should imply a smaller pass-through.	The higher the staff costs, the smaller the impact of monetary policy shocks on bank credit rates.

### Table 3 (continued). Potential Determinants to Interest Rate Pass-Through Analyzed in the Literature

Sources	Factors	Variable	Impact on transmission	Empirical findings
Mojon (2000), Sørensen and Werner (2006)	Income structure	Ratios of banks' non- interest income to banks' gross income.	A higher share of non-interest income to total gross income seems to speed up banks' rate adjustment. Banks that are relatively less dependent on interest-related income adjust their interest rates more quickly to a change in the market rate, perhaps to capture market shares in a competitive environment. It may be expected that a bank with a highly diversified portfolio of activities (i.e. banks that do not only rely on traditional banking activities such as granting loans and taking deposits, e.g. measured by the share of non-interest income) may be less sensitive to movements in market rates, which would imply a more sluggish pass-through. At the same time, it cannot be ruled out that in a highly competitive environment, very diversified banks may be able to exploit this by offering more attractive rates to conquer market shares, implying a speedier pass-through.	Results are mixed.
Cottarelli and Kourelis (1994), Mojon (2000)	Competition from direct finance	Size of the market for short-term negotiable financial instruments issued by enterprises and other agents, both measured in relation to each country's GDP; The ratio of commercial paper and total short- term securities to GDP.	The existence of a market for short-term instruments issued by enterprises (commercial paper and bankers' acceptances) may be relevant because it increases the elasticity of the demand for bank loans. In this case, if banks do not adjust rapidly to changes in money market conditions, they may be disintermediated. The existence of a market for other short-term marketable instruments (mainly certificates of deposit (CDs), and Treasury bills) may also be important. The existence of these instruments increases the liquidity of enterprise and household portfolios, thus increasing the elasticity of demand for loans. Moreover, if banks raise a large share of their resources from the issuance of CDs, whose interest rates rapidly adjust to money market conditions, they will face large costs if they delay the adjustment of their lending rates.	The development of a market for short-term instruments enhances the flexibility of lending rates.
Cottarelli and Kourelis (1994)	Openness of an economy	The barriers to foreign competition.	Capital controls reduce competitive pressures on the banking system (arising from foreign financial markets) and result in higher lending rate stickiness.	Negative effect on pass- through.
Cottarelli and Kourelis (1994)	Banking system ownership	Number of public banks out of the five largest ones.	The relative inefficiency of public banks or the existence of "political constraints" on interest rate changes make lending rates stickier.	Lending rates appear to be stickier in publicly- owned banking systems. Privatizing a publicly owned banking system would substantially increase the flexibility of lending rates.

### Table 3 (continued). Potential Determinants to Interest Rate Pass-Through Analyzed in the Literature

Sourcos	Eactors	Variable	Impact on transmission	Empirical findings
Sources Sørensen and Werner (2006)	Interest rate risk (maturity mismatch)	LT assets/total assets vs. LT liabilities/total liabilities	The higher the interest rate risk banks are exposed to, the more they need to hedge and through their hedging activities the more sensitive they are to changes in market rates. This would tend to imply a speedier pass-through. Alternatively, it may be argued that the interest rate risk is linked to the degree of capitalization in the sense that the profit (and hence capital accumulation) of banks having a relatively large maturity mismatch is more sensitive to changes in market rates, which could induce them to adjust their interest rates more slowly to compensate for this potential loss.	Negative effect on pass- through.
Stanisławska (2015), Sørensen and Werner (2006)	Structure of funding	Share of non-financial sector deposits in liabilities	Banks with a large and stable pool of deposit funding (e.g. measured by the share of deposits to total liabilities) would be expected to be less vulnerable to changes in market rates (as most of their funding is non-market based) thereby leading to a relatively slower speed of adjustment. The rigidity of funding costs depends mainly on pricing practices in the banking sector and on the extent to which the interest rate received or paid by banks is itself rigid. A bank that can rely on traditional deposits is more likely to have more rigid funding costs than a bank that funds itself mostly by issuing debt on the capital markets	Negative effect on pass- through.

# Table 3 (continued). Potential Determinants to Interest Rate Pass-Through Analyzed in the Literature

Table 4. Valiables, L	Jenniuons, and Sources			
Notation	Variable	Description	Level	Source
HH_TR, NFCL_TR	Long-run transmission to deposit (HH) and credit (NFCL) rates	Long–run coefficient of transmission estimated via some modification of ARDL (linear, nonlinear, TVP)	Economy, Banks	authors' calculations
HHD_R	HH deposit rates	Weighted average interest rates on term deposits of HHs in national currency	Economy, Banks	NBU
NFCL_R	NFC credit rates	Weighted average interest rates on credits to NFCs in national currency	Economy, Banks	NBU
UONIA	Interbank overnight rate	Ukrainian weighted average overnight interbank rate	Economy	NBU
HHD_SHR	Share of deposits	Share of term deposits of HHs in national currency in the current month	Banks	NBU
NFCL_SHR	Share of credits	Share of credits to NFCs in national currency in the current month	Banks	NBU
NPL_AGR	Share of non-performing loans	Non-performing loans, mln.hrn./ Total loans to residents and households, mln. hrn.	Banks	NBU
FU_CST_HH	Cost of funds acquired from HHs	Payments to HHs, mln.hrn. / Funds attracted from HHs, mln.hrn.	Banks	NBU
FU_CST_NFC	Cost of funds acquired from NFCLs	Payments to NFCs, mln.hrn. / Funds attracted from NFCs mln.hrn.	Banks	NBU
P_VK	Profit after taxation to equity capital ratio	Profit (loss) after tax, mln.hrn. / Total equity capital, mln.hrn.	Banks	NBU, Banks' balance sheet reports
INTREV_A	Net interest revenues to assets ratio	Net interest margin, mln.hrn. / Total assets, mln.hrn.	Banks	NBU, Banks' balance sheet reports
RES_A	Provisions to assets ratio	-1*(Total provisions, mln.hrn. / Total assets, mln.hrn.)	Banks	NBU, Banks' balance sheet reports
VK_LIAB	Equity capital to liabilities ratio	Total equity capital, mln.hrn. / Total liabilities, mln.hrn.	Banks	NBU, Banks' balance sheet reports
SHORTSEC_A	Short-term securities of the NBU to assets	Securities at amortized cost (including refinanced by NBU), mln.hrn. / Total assets, mln.hrn	Banks	NBU, Banks' balance sheet reports
OVDP_A	Government bonds to assets	IGLB refinanced by NBU, mln.hrn. / Total assets, mln.hrn. IGLB – domestic bonds denominated in national currency	Banks	NBU, Banks' balance sheet reports
LIQ_A	Share of liquid assets	Cash and cash equivalents, mln.hrn. / Total assets, mln.hrn	Banks	NBU, Banks' balance sheet reports
TERMDEP_LIAB	Share of term deposits in liabilities	(Amounts due to customers – demand deposits from legal entities – demand deposits from individuals), mln.hrn. / Total liabilities, mln.hrn.	Banks	NBU, Banks' balance sheet reports
SEC_LIAB	Share of securities in total liabilities	(Debt securities issued by the bank + Other funds raised), mln.hrn. / Total liabilities payroll costs – tax on payroll – other staff costs	Banks	NBU, Banks' balance sheet reports
INTREV_REV	Share of interest income	Interest income, mln.hrn. / Gross income, mln.hrn.	Banks	NBU, Banks' balance sheet reports
NADMEXP_EXP	Administrative expenses to gross expenses ratio	(Administrative and other operating expenses – payroll costs – tax on payroll – other staff costs), mln.hrn. / Gross expenditures, mln.hrn.	Banks	NBU, Banks' balance sheet reports
PERSEXP_EXP	Staff costs to gross expenses ratio	(Payroll costs + tax on payroll + other staff costs), mln.hrn. / Gross expenditures, mln. hrn.	Banks	NBU, Banks' balance sheet reports
INTEXP_EXP	Share of interest expenses	Interest expenses, mln.hrn. / Gross expenses, mln.hrn.	Banks	NBU, Banks' balance sheet reports
LIQ	Liquidity in the banking	CDs on banks' balance sheets, mln.hrn. / Total liabilities of banks, mln.hrn.	Economy	NBU

### Table 4. Variables, Definitions, and Sources

Notation	Variable	Description	Level	Source
FSI	Financial stress index	The Financial Stress Index (FSI) is an indicator representing the stress level in the financial sector of Ukraine. It fluctuates from 0 to 1, where 0 – total absence of stress and 1 is the highest level of stress.	Economy	NBU
HHI_DEP	Herfindahl-Hirschman index for deposit market	Index estimated as a sum of squared market shares of each bank in banking system. It demonstrates concentration of HH deposit market. The index fluctuates from 0 to 1, increase means higher market concentration.	Economy	NBU, authors' calculations
HHI_CRED	Herfindahl-Hirschman index for credit market	Index estimated as a sum of squared market shares of each bank in banking system. It demonstrates concentration of NFC credit market. The index fluctuates from 0 to 1, increase means higher market concentration.	Economy	NBU, authors' calculations
CPI	Inflation	Year-over-year CPI change, %	Economy	NBU
RDEP_GR	Real growth of deposits	Proxy for deposits supply. Year-over-year growth of HH deposits deflated by CPI.	Economy	NBU, authors' calculations
RCRED_GR	Real growth of credits	Proxy for credits demand. Year-over-year growth of NFC credits deflated by CPI.	Economy	NBU, authors' calculations
DC_100D_SHR	Long-term NBU's certificates of deposit (CDs)	The share of CDs with a maturity of more than one day in the total amount of CDs allocated to banks.	Economy	NBU
NORM_RES	Reserve requirement ratio for HH deposits on demand and current accounts	Effective reserve requirement ratio for demand deposits and deposits on current accounts of individuals.	Economy	NBU

### Table 4 (continued). Variables, Definitions, and Sources

Variable	Mean	Median	Max.	Min.	SD	Obs.	Banks	Sample period
UONIA	13.68	15.76	23.01	5.08	5.97	72	Economy	2018m1-2023m12
NFCL_R	18.62	19.27	38.04	3.46	4.35	3905	55	2018m1-2023m12
HHD_R	11.56	12.21	30.37	0.03	3.59	1800	25	2018m1-2023m12
HH_TR	0.72	0.72	1.09	0.38	0.12	1800	25	2018m1-2023m12
NFCL_TR	1.14	1.14	1.94	0.45	0.20	3905	55	2018m1-2023m12
NFCL_SHR	0.02	0.00	0.44	0.00	0.05	3905	55	2018m1-2023m12
HHD_SHR	0.04	0.01	0.43	0.00	0.08	1800	25	2018m1-2023m12
P_VK	0.01	0.01	0.46	-1.23	0.05	3905	55	2018m1-2023m12
INTREV_A	0.01	0.00	0.04	-0.02	0.00	3905	55	2018m1-2023m12
NPL_AGR	0.20	0.15	0.85	0.00	0.18	3905	55	2018m1-2023m12
RES_A	0.07	0.05	0.47	0.00	0.08	3905	55	2018m1-2023m12
FU_CST_HH	0.05	0.05	0.34	0.00	0.03	3905	55	2018m1-2023m12
FU_CST_NFC	0.05	0.04	0.25	0.00	0.03	3905	55	2018m1-2023m12
VK_LIAB	0.35	0.17	15.48	0.01	0.75	3905	55	2018m1-2023m12
SHORTSEC_A	0.15	0.11	0.75	0.00	0.14	3905	55	2018m1-2023m12
OVDP_A	0.13	0.09	0.74	0.00	0.14	3905	55	2018m1-2023m12
LIQ_A	0.08	0.06	0.39	0.00	0.05	3905	55	2018m1-2023m12
TERMDEP_LIAB	0.35	0.34	0.87	0.00	0.19	3905	55	2018m1-2023m12
SEC_LIAB	0.02	0.00	0.42	0.00	0.07	3905	55	2018m1-2023m12
NADMEXP_EXP	0.23	0.22	0.88	-0.73	0.11	3905	55	2018m1-2023m12
PERSEXP_EXP	0.25	0.24	0.87	0.00	0.12	3905	55	2018m1-2023m12
INTREV_REV	0.63	0.66	1.63	-0.42	0.18	3905	55	2018m1-2023m12
INTEXP_EXP	0.30	0.28	0.83	0.00	0.15	3905	55	2018m1-2023m12
LIQ	0.06	0.04	0.19	0.01	0.04	108	Economy	2015m1-2023m12
FSI	0.14	0.04	0.84	0.00	0.18	108	Economy	2015m1-2023m12
HHI_CRED	0.16	0.17	0.22	0.08	0.03	108	Economy	2015m1-2023m12
HHI_DEP	0.22	0.22	0.23	0.18	0.01	108	Economy	2015m1-2023m12
CPI	16	11	61	2	0.14	108	Economy	2015m1-2023m12
RCRED_GR	-13	-9	10	-60	0.17	108	Economy	2015m1-2023m12
RDEP_GR	-11	-3	34	-78	0.24	108	Economy	2015m1-2023m12
DC_100D_SHR	0.64	0.74	0.93	0	0.28	108	Economy	2015m1-2023m12
NORM RES	5.68	6.5	20	0	5.35	108	Economy	2015m1-2023m12

 Table 5. Descriptive Statistics of Variables

Bank, NKB	Optimal Model	R <sup>2</sup>	Test for serial correlation <sup>11</sup>	Test for heteroscedasticity <sup>12</sup>	F bonds test	1 – residuals are not serially correlated	1 – homoscedasticity of residuals	1 – long-run relationship identified at 10% probability
2	ARDL(1, 0)	0.97	0.85	0.16	15.77	1	1	1
6	ARDL(1, 0)	0.96	0.78	0.36	8.13	1	1	1
36	ARDL(1, 5)	0.87	0.02	0.00	3.49	0	0	0
46	ARDL(1, 0)	0.98	0.23	0.02	7.91	1	0	1
88	ARDL(3, 4)	0.80	0.32	0.34	1.75	1	1	0
115	ARDL(1, 1)	0.93	0.42	0.34	3.81	1	1	1
136	ARDL(2, 0)	0.54	0.16	0.65	6.99	1	1	1
274	ARDL(2, 1)	0.95	0.19	0.00	5.86	1	0	1
49	ARDL(1, 0)	0.89	0.30	0.32	12.35	1	1	1
62	ARDL(1, 1)	0.90	0.60	0.81	4.41	1	1	1
91	ARDL(3, 0)	0.76	0.30	0.11	8.09	1	1	1
96	ARDL(2, 0)	0.93	0.38	0.83	7.25	1	1	1
105	ARDL(2, 0)	0.94	0.38	0.00	8.53	1	0	1
106	ARDL(2, 0)	0.80	0.15	0.14	9.68	1	1	1
123	ARDL(1, 0)	0.89	0.04	0.01	3.09	0	0	0
142	ARDL(3, 0)	0.91	0.25	0.76	7.82	1	1	1
153	ARDL(2, 3)	0.85	0.14	0.01	10.44	1	0	1
171	ARDL(1, 0)	0.87	0.09	0.21	4.72	0	1	1
205	ARDL(1, 0)	0.87	0.44	0.11	7.06	1	1	1
242	ARDL(1, 0)	0.91	0.20	0.21	3.09	1	1	0
270	ARDL(1, 0)	0.89	0.67	0.17	11.85	1	1	1
272	ARDL(2, 6)	0.94	0.09	0.00	1.61	0	0	0
290	ARDL(3, 0)	0.62	0.01	0.86	6.78	0	1	1
296	ARDL(2, 1)	0.71	0.09	0.87	11.44	0	1	1
320	ARDL(1, 0)	0.85	0.28	0.73	5.87	1	1	1

 $\textbf{Table 6.} \ \text{Information on ARDL Models for Deposits}^*$ 

\*HAC covariance matrix estimator is used to address issues of heteroscedasticity and serial correlation in regressions

Bank code	Optimal Model	timal R <sup>2</sup> Test for serial Test for correlation <sup>13</sup> heteroscedasticity <sup>14</sup>		Test for heteroscedasticity <sup>14</sup>	F bonds test	1 – residuals are not serially correlated	1 – homoscedasticity of residuals	1 – long-run relationship identified
2	ARDL(3, 1)	0.94	0.20	0.99	23.65	1	1	1
6	ARDL(3, 0)	0.76	0.04	0.08	6.19	0	0	1
29	ARDL(1, 5)	0.69	0.28	0.00	15.13	1	0	1
36	ARDL(1, 0)	0.95	0.49	0.00	17.82	1	0	1
43	ARDL(4, 0)	0.84	0.01	0.03	4.14	0	0	1
46	ARDL(1, 0)	0.68	0.91	0.18	5.39	1	1	1
49	ARDL(2, 2)	0.81	0.30	0.00	14.39	1	0	1
62	ARDL(3, 0)	0.91	0.71	0.67	9.89	1	1	1
88	ARDL(1, 0)	0.98	0.67	0.06	24.05	1	0	1
91	ARDL(3, 0)	0.86	0.55	0.23	11.37	1	1	1
95	ARDL(2, 0)	0.51	0.00	0.50	5.74	0	1	1
96	ARDL(2, 6)	0.60	0.08	0.15	3.76	0	1	1
101	ARDL(2, 1)	0.76	0.02	0.00	4.21	0	0	1
105	ARDL(2, 0)	0.91	0.83	0.44	12.22	1	1	1
106	ARDL(3, 0)	0.81	0.83	0.00	9.24	1	0	1
113	ARDL(3, 0)	0.95	0.29	0.05	9.12	1	0	1
115	ARDL(2, 0)	0.86	0.37	0.00	7.22	1	0	1
123	ARDL(1, 4)	0.93	0.13	0.04	1.49	1	0	0
128	ARDL(3, 0)	0.55	0.17	0.72	4.28	1	1	1
133	ARDL(3, 0)	0.62	0.18	0.00	4.43	1	0	1
136	ARDL(1, 2)	0.96	0.20	0.14	9.19	1	1	1
143	ARDL(2, 0)	0.80	0.13	0.02	1.37	1	0	0
146	ARDL(2, 5)	0.93	0.28	0.18	4.82	1	1	1
153	ARDL(2, 0)	0.92	0.83	0.00	17.65	1	0	1
171	ARDL(2, 4)	0.96	0.04	0.00	8.69	0	0	1
205	ARDL(2, 5)	0.86	0.00	0.00	8.01	0	0	1
231	ARDL(2, 0)	0.58	0.01	0.62	5.20	0	1	1
240	ARDL(2, 0)	0.80	0.45	0.24	7.96	1	1	1
242	ARDL(3, 0)	0.16	0.73	0.02	6.23	1	0	1
251	ARDL(5, 0)	0.85	0.00	0.16	7.32	0	1	1
270	ARDL(2, 0)	0.62	0.71	0.01	5.00	1	0	1
272	ARDL(2, 0)	0.76	0.10	0.04	14.11	1	0	1
274	ARDL(3, 0)	0.91	0.73	0.25	12.66	1	1	1
286	ARDL(5, 6)	0.88	0.49	0.00	3.28	1	0	0
288	ARDL(3, 0)	0.74	0.87	0.05	1.09	1	0	0
295	ARDL(4, 4)	0.98	0.18	0.00	1.56	1	0	0
296	ARDL(5, 5)	0.95	0.62	0.00	3.67	1	0	1
297	ARDL(1, 1)	0.93	0.33	0.07	19.96	1	0	1
298	ARDL(3, 0)	0.96	0.40	0.00	19.49	1	0	1
305	ARDL(3, 0)	0.92	0.01	0.52	17.19	0	1	1
320	ARDL(3, 0)	0.72	0.18	0.03	12.31	1	0	1
331	ARDL(3, 0)	0.90	0.10	0.00	4.93	1	0	1
377	ARDL(4, 0)	0.63	0.18	0.00	6.33	1	0	1
381	ARDL(2, 0)	0.89	0.41	0.37	10.69	1	1	1

 $\textbf{Table 7.} \ \text{Information on ARDL Models for Loans}^*$ 

<sup>13</sup> Breusch-Godfrey Lagrange Multiplier test for serial correlation

<sup>14</sup> Breusch-Pagan-Godfrey test for heteroscedasticity.

Bank code	Optimal Model	R <sup>2</sup>	Test for serial correlation <sup>13</sup>	Test for heteroscedasticity <sup>14</sup>	F bonds test	1 – residuals are not serially correlated	1 – homoscedasticity of residuals	1 – long-run relationship identified
386	ARDL(2, 2)	0.84	0.29	0.00	4.55	1	0	1
387	ARDL(2, 0)	0.61	0.01	0.44	5.46	0	1	1
389	ARDL(4, 0)	0.89	0.57	0.01	11.50	1	0	1
392	ARDL(1, 0)	0.44	0.34	0.42	29.56	1	1	1
394	ARDL(3, 0)	3, 0) 0.27 0.32	0.32	0.08	4.50	1	0	1
395	ARDL(3, 0)	0.41	0.25	0.52	3.58	1	1	1
407	ARDL(2, 2)	0.96	0.42	0.62	5.19	1	1	1
455	ARDL(2, 4)	0.91	0.06	0.88	3.66	0	1	1
634	ARDL(1, 0)	0.45	0.19	0.31	6.99	1	1	1
694	ARDL(1, 0)	0.75	0.00	0.10	7.98	0	1	1
774	ARDL(1, 0)	0.85	0.16	0.91	2.62	1	1	0

|--|

\* HAC covariance matrix estimator is used to address issues of heteroscedasticity and serial correlation in regressions

		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
	HHD_SHR	1.335	1.335	1.335**				1.89	1.89	1.89***
	P_VK	-0.001	-0.001	-0.001				-0.017	-0.017	-0.017
	INTREV_A	-2.132	-2.132	-2.132				-1.675	-1.675	-1.675
	NPL_AGR	-0.056	-0.056	-0.056**				-0.033	-0.033	-0.033
	RES_A	0.33*	0.33*	0.33***				0.359**	0.359**	0.359***
oles	FU_CST_HHD	-0.100	-0.100	-0.100				0.767*	0.767	0.767***
ariak	FU_CST_NFC	-0.389	-0.389	-0.389**				-0.172	-0.172	-0.172
k-level va	VK_LIAB	-0.034	-0.034	-0.034				-0.010	-0.010	-0.010
	SHORTSEC_A	-0.033	-0.033	-0.033				-0.008	-0.008	-0.008
Ban	LIQ_A	-0.059	-0.059	-0.059				0.028	0.028	0.028
	TERMDEP_LIAB	0.027	0.027	0.027				0.014	0.014	0.014
	NADMEXP_EXP	0.006	0.006	0.006				0.019	0.019	0.019
	PERSEXP_EXP	0.091	0.091	0.091***				0.071	0.071	0.071**
	INTEXP_EXP	0.014	0.014	0.014				0.032	0.032	0.032
	INTREV_REV	0.056*	0.056*	0.056***				0.031**	0.031*	0.031**
	LIQ				-0.807***	-0.807***	-0.807***	-0.586***	-0.586***	-0.586***
S	FSI				-0.086***	-0.086*	-0.086***	-0.083***	-0.083***	-0.083***
iable	HHI_DEP				1.261***	1.261***	1.261***	0.77**	0.77*	0.77***
var	CPI				-0.623***	-0.623**	-0.623***	-0.605***	-0.605***	-0.605***
acro	RDEP_GR				-0.243***	-0.243*	-0.243***	-0.232***	-0.232***	-0.232***
Σ	DC_100D_SHR				0.073***	0.073**	0.073***	0.043**	0.043*	0.043**
	NORM_RES				0.004***	0.004**	0.004***	0.003*	0.003*	0.003***
	Bank FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
	Time FE	Yes	Yes	Yes	No	No	No	No	No	No
	Heteroscedasticity	One-way clusterization	Two-way clusterization	HAC	One-way clusterization	Two-way clusterization	HAC	One-way clusterization	Two-way clusterization	HAC
	Autocorrelation			HAC			HAC			HAC
	Ν	1,800	1,800	1,800	1,800	1,800	1,800	1,800	1,800	1,800
	Within R <sup>2</sup>	0.735	0.891	0.147	0.633	0.849	0.633	0.696	0.875	0.696

		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
	NFCL_SHR	0.162	0.162	0.162				0.654	0.654	0.654**
	P_VK	-0.039	-0.039	-0.039				-0.017	-0.017	-0.017
	INTREV_A	7.486**	7.486**	7.486***				4.426	4.426	4.426**
	NPL_AGR	-0.045	-0.045	-0.045				-0.089	-0.089	-0.0889**
	RES_A	0.401	0.401	0.401***				0.266	0.266	0.266**
S	FU_CST_HHD	-0.023	-0.023	-0.023				-0.400	-0.400	-0.400*
iable	FU_CST_NFC	0.269	0.269	0.269				-0.254	-0.254	-0.254
var	VK_LIAB	0.014*	0.014	0.014				0.006	0.006	0.006
evel	SHORTSEC_A	0.012	0.012	0.012				0.025	0.025	0.025
l-yu	OVDP_A	-0.010	-0.010	-0.010				0.056	0.056	0.0564*
ä	LIQ_A	0.200	0.200	0.200**				0.059	0.059	0.059
	TERMDEP_LIAB	-0.062	-0.062	-0.062*				-0.108	-0.108	-0.108***
	NADMEXP_EXP	0.004	0.004	0.004				-0.005	-0.005	-0.005
	PERSEXP_EXP	-0.007	-0.007	-0.007				0.011	0.011	0.011
	INTEXP_EXP	0.075	0.075	0.0753**				0.019	0.019	0.019
	INTREV_REV	-0.029	-0.029	-0.029				-0.014	-0.014	-0.014
	LIQ				-0.242*	-0.242	-0.242***	-0.344*	-0.344*	-0.344***
oles	FSI				0.075***	0.075*	0.075**	0.047*	0.047*	0.047
ariak	HHI_CRED				0.787**	0.787	0.787**	0.987*	0.987*	0.987***
ro v	CPI				-0.150	-0.150	-0.150	-0.166	-0.166	-0.166
Mag	RCRED_GR				0.256***	0.256**	0.256***	0.169**	0.169*	0.169**
	DC_100D_SHR				0.066***	0.066	0.066***	0.061***	0.061*	0.061***
	Bank FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
	Time FE	Yes	Yes	Yes	No	No	No	No	No	No
	Heteroscedasticity	One-way clusterization	Two-way clusterization	HAC	One-way clusterization	Two-way clusterization	HAC	One-way clusterization	Two-way clusterization	HAC
	Autocorrelation			HAC			HAC			HAC
	Ν	3,960	3,960	3,960	3,960	3,960	3,960	3,960	3,960	3,960
	Within R <sup>2</sup>	0.343	0.819	0.044	0.237	0.790	0.237	0.281	0.802	0.281

 Table 9. Panel Regression: Determinants of Transmission to Credit Rates, (2018m1 – 2023m12)

		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	Transmission horizon	31	im 61		m	91	m	12	m
	HHD_SHR	0.410	0.410***	0.860	0.860***	1.240	1.240***	1.450	1.450***
	P_VK	-0.010	-0.013*	-0.020	-0.020	-0.020	-0.020	-0.020	-0.020
	INTREV_A	-0.720	-0.720*	-1.200	-1.202*	-1.510	-1.513*	-1.710	-1.710*
	NPL_AGR	0.000	0.000	0.000	0.000	-0.010	-0.010	-0.020	-0.020
	RES_A	0.060	0.060**	0.120	0.120***	0.170*	0.170***	0.200*	0.200***
oles	FU_CST_HHD	0.120	0.115*	0.220	0.218*	0.300	0.300*	0.390	0.390**
arial	FU_CST_NFC	-0.050	-0.050	-0.100	-0.100	-0.140	-0.140	-0.170	-0.170
/el v	VK_LIAB	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
k-lev	SHORTSEC_A	-0.020	-0.020*	-0.020	-0.020	-0.010	-0.010	-0.010	-0.010
Ban	LIQ_A	0.050	0.050	0.080	0.080	0.090	0.090	0.090	0.090
	TERMDEP_LIAB	-0.020	-0.020	-0.020	-0.020	-0.020	-0.020	-0.010	-0.010
	NADMEXP_EXP	0.020	0.020**	0.020	0.020*	0.020	0.020*	0.020	0.020
	PERSEXP_EXP	0.010	0.010	0.020	0.020	0.030	0.030	0.040	0.040
	INTEXP_EXP	0.010	0.010	0.030	0.030	0.030	0.030	0.030	0.030
	INTREV_REV	0.000	0.000	0.000	0.000	0.010	0.010	0.010	0.010
	LIQ	-0.300***	-0.300***	-0.500***	-0.500***	-0.540***	-0.540***	-0.580***	-0.580***
S	FSI	-0.020*	-0.020*	-0.040*	-0.040**	-0.050**	-0.050**	-0.060**	-0.060**
iable	HHI_DEP	0.300*	0.300***	0.500*	0.500***	0.620**	0.620***	0.690**	0.690**
var	CPI	-0.120	-0.120**	-0.250*	-0.250***	-0.350**	-0.350***	-0.420**	-0.420**
acro	RDEP_GR	-0.050	-0.050*	-0.100*	-0.100**	-0.130*	-0.130**	-0.160**	-0.160**
Σ	DC_100D_SHR	0.020**	0.020***	0.035**	0.035***	0.042**	0.042***	0.045**	0.045**
	NORM_RES	0.001**	0.001***	0.002**	0.002***	0.002**	0.002***	0.002**	0.002**
	Bank FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
	Time FE	No	No	No	No	No	No	No	No
	Heteroscedasticity	Two-way clusterization	HAC						
	Autocorrelation		HAC		HAC		HAC		HAC
	N	1,800	1,800	1,800	1,800	1,800	1,800	1,800	1,800
	Within R <sup>2</sup>	0.990	0.700	0.960	0.710	0.940	0.710	0.920	0.710

Table 10. Panel Regression: Determinants of Transmission to Deposit Rates for Different Horizons, (2018m1 – 2023m12)

		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	Transmission horizon	3	m	6	m	9	m	12	m
	NFCL_SHR	0.150	0.150**	0.270	0.270**	0.350	0.350***	0.400	0.400***
	P_VK	0.000	0.000	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010
	INTREV_A	0.240	0.240	0.700	0.700	1.460	1.460*	2.080	2.080**
	NPL_AGR	-0.010	-0.010	-0.010	-0.010	-0.020	-0.020*	-0.030	-0.030*
	RES_A	0.000	0.000	0.020	0.020	0.030	0.030	0.050	0.050
S	FU_CST_HHD	-0.010	-0.010	0.020	0.020	0.020	0.020	0.000	0.000
iable	FU_CST_NFC	0.060	0.060**	0.040	0.040	-0.010	-0.010	-0.060	-0.060
l var	VK_LIAB	0.004***	0.004**	0.006**	0.006*	0.006*	0.006*	0.010	0.010
eve	SHORTSEC_A	-0.010	-0.010**	-0.020	-0.020*	-0.020	-0.020	-0.020	-0.020
-yue	OVDP_A	0.010	0.010**	0.030	0.030***	0.040	0.040***	0.040	0.050***
ä	LIQ_A	0.010	0.010	0.020	0.020	0.020	0.020	0.030	0.030
	TERMDEP_LIAB	-0.030*	-0.030*	-0.050*	-0.050***	-0.050	-0.050***	-0.060	-0.060
	NADMEXP_EXP	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010
	PERSEXP_EXP	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
	INTEXP_EXP	0.010	0.010**	0.010	0.010	0.010	0.010	0.020	0.020
	INTREV_REV	0.000	0.000	0.000	0.000	-0.010	-0.010	-0.010	-0.010
	LIQ	-0.080***	-0.080***	-0.130**	-0.130***	-0.17**	-0.170***	-0.190**	-0.190***
oles	FSI	0.000	0.000	0.000	0.000	0.010	0.010	0.010*	0.010
arial	HHI_CRED	0.150***	0.150***	0.270**	0.270***	0.370**	0.370***	0.460**	0.460***
ro v	CPI	0.002	0.002	-0.010	-0.010	-0.020	-0.020	-0.030	-0.030
Mag	RCRED_GR	0.026***	0.026**	0.048**	0.048**	0.068**	0.068**	0.084**	0.084**
	DC_100D_SHR	0.007***	0.007**	0.012**	0.012**	0.018**	0.018**	0.023*	0.023**
	Bank FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
	Time FE	No	No	No	No	No	No	No	No
	Heteroscedasticity	Two-way clusterization	HAC						
	Autocorrelation		HAC		HAC		HAC		HAC
	Ν	3,960	3,960	3,960	3,960	3,960	3,960	3,960	3,960
	Within R <sup>2</sup>	0.997	0.293	0.992	0.303	0.983	0.303	0.973	0.302

		(1)	(2)	(3)	(4)	(5)	(6)
	HHD_SHR	2.050	2.050	2.048***	1.700	1.700	1.699***
	P_VK	-0.020	-0.020	-0.020	-0.020	-0.020	-0.020
	INTREV_A	-1.620	-1.620	-1.620	-1.470	-1.470	-1.470
	NPL_AGR	-0.020	-0.020	-0.020	-0.040	-0.040	-0.043*
	RES_A	0.353**	0.353**	0.353***	0.346**	0.346**	0.346***
oles	FU_CST_HHD	1.037*	1.037*	1.037***	0.540	0.540	0.536**
ariat	FU_CST_NFC	-0.230	-0.230	-0.230	-0.230	-0.230	-0.230
/el v	VK_LIAB	0.020	0.020	0.020	-0.020	-0.020	-0.020
k-lev	SHORTSEC_A	-0.090	-0.090	-0.086**	-0.020	-0.020	-0.020
Ban	LIQ_A	-0.060	-0.060	-0.060	-0.020	-0.020	-0.020
	TERMDEP_LIAB	0.040	0.040	0.040	0.020	0.020	0.020
	NADMEXP_EXP	0.010	0.010	0.010	0.010	0.010	0.010
	PERSEXP_EXP	0.050	0.050	0.050	0.080	0.080	0.076**
	INTEXP_EXP	0.030	0.030	0.030	0.030	0.030	0.030
	INTREV_REV	0.027*	0.027*	0.027*	0.029*	0.029*	0.029**
	LIQ				-0.731***	-0.731***	-0.731***
	FSI	-0.107***	-0.107**	-0.107***	-0.077***	-0.077**	-0.077***
oles	HHI_DEP	0.220	0.220	0.220	0.02	0.020	0.020
arial	CPI	-1.020***	-1.020***	-1.020***	-1.260***	-1.260***	-1.260***
ro v	CPI <sup>2</sup>				2.624**	2.624**	2.624***
Mag	RDEP_GR	-0.502***	-0.502***	-0.502***	-0.250***	-0.250***	-0.250***
	DC_100D_SHR	0.059***	0.059**	0.059***	0.049**	0.049**	0.049***
	NORM_RES	0.004**	0.004**	0.004***	0.005**	0.005**	0.005***
	Bank FE	Yes	Yes	Yes	Yes	Yes	Yes
	Time FE	No	No	No	No	No	No
	Heteroscedasticity	One-way clusterization	Two-way clusterization	HAC	One-way clusterization	Two-way clusterization	HAC
	Autocorrelation			HAC			HAC
	Ν	1,800	1,800	1,800	1,800	1,800	1,800
	Within R <sup>2</sup>	0.671	0.864	0.671	0.701	0.876	0.701

Table 12.	Alternative	Specifications	of Regressions for	Long-Run	n Transmission to Deposit Rates, 2018m1–20	023m12
-----------	-------------	----------------	--------------------	----------	--	--------

		(1)	(2)	(3)	(4)	(5)	(6)
	NFCL_SHR	0.540	0.540	0.542*	0.600	0.600	0.600*
	P_VK	-0.020	-0.020	-0.020	-0.020	-0.020	-0.020
	INTREV_A	4.700	4.700	4.695**	4.980	4.980	4.970**
	NPL_AGR	-0.100	-0.1000	-0.102***	-0.080	-0.080	-0.080**
	RES_A	0.290	0.290	0.287***	0.280	0.280	0.280**
S	FU_CST_HHD	-0.310	-0.310	-0.310	-0.340	-0.340	-0.335*
iable	FU_CST_NFC	-0.220	-0.220	-0.220	-0.170	-0.170	-0.170
l var	VK_LIAB	0.010	0.010	0.010	0.010	0.010	0.010
eve	SHORTSEC_A	-0.010	-0.010	-0.010	0.020	0.020	0.020
ank-l	OVDP_A	0.050	0.050	0.050	0.060	0.060	0.060*
Bâ	LIQ_A	0.000	0.000	0.000	0.090	0.090	0.090
	TERMDEP_LIAB	-0.100	-0.100	-0.103***	-0.100	-0.100	-0.100**
	NADMEXP_EXP	0.000	0.000	0.000	0.000	0.000	0.000
	PERSEXP_EXP	0.020	0.020	0.020	0.010	0.010	0.010
	INTEXP_EXP	0.010	0.010	0.010	0.0300	0.0300	0.030
	INTREV_REV	-0.020	-0.020	-0.020	-0.02	-0.02	-0.020
	LIQ				-0.494***	-0.494**	-0.494***
S	FSI	0.046*	0.050	0.050	0.052**	0.052*	0.0521*
iable	HHI_CRED	1.439**	1.439*	1.439***	-0.010	-0.010	-0.010
o var	CPI	0.000	0.000	0.000	-1.097***	-1.097**	-1.097***
acro	CPI <sup>2</sup>				2.788**	2.788**	2.788***
Σ	RCRED_GR	0.314**	0.314**	0.314***	0.080	0.080	0.080
	DC_100D_SHR	0.081***	0.081*	0.081***	0.078***	0.078**	0.078***
	Bank FE	Yes	Yes	Yes	Yes	Yes	Yes
	Time FE	No	No	No	No	No	No
	Heteroscedasticity	One-way clusterization	Two-way clusterization	HAC	One-way clusterization	Two-way clusterization	HAC
	Autocorrelation			HAC			HAC
	Ν	3,960	3,960	3,960	3,960	3,960	3,960
	Within R <sup>2</sup>	0.273	0.800	0.273	0.289	0.804	0.289

Table 13. Alternative Specifications of Regressions for Long-Ru	n Transmission to Credit Rates, 2018m1–2023m12
---	--

		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
	Transmission horizon	3m		6m		9m		12m		LR	
	HHD_SHR	0.330	0.330**	0.740	0.740***	1.090	1.090***	1.280	1.280***	1.690	1.690***
	P_VK	0.001	0.001	0.001	0.001	0.001	0.001	0.001	0.001	0.001	0.001
	INTREV_A	-0.220	-0.220	-0.650	-0.650	-0.980	-0.980	-1.160	-1.160	-0.800	-0.800
	NPL_AGR	0.020	0.020*	0.020	0.020	0.010	0.010	0.000	0.000	-0.020	-0.020
	RES_A	0.020	0.020	0.060	0.060	0.090	0.090*	0.120	0.120**	0.210	0.210**
	FU_CST_HH	0.110	0.110	0.220	0.220*	0.350	0.350**	0.490	0.490**	1.100**	1.100***
S	FU_CST_NFCL	-0.040	-0.040	-0.050	-0.050	-0.040	-0.040	-0.050	-0.050	0.000	0.000
iable	VK_LIAB	-0.040*	-0.040***	-0.060	-0.060***	-0.070	-0.070***	-0.080	-0.070***	-0.120*	-0.120***
l var	SHORTSEC_A	-0.030	-0.030**	-0.030	-0.030	-0.020	-0.020	-0.010	-0.010	-0.010	-0.010
eve	LIQ_A	0.050	0.050*	0.100	0.100*	0.130	0.130*	0.140	0.140*	0.120	0.120
-yue	TERMDEP_LIAB	-0.040	-0.040***	-0.050	-0.050**	-0.060	-0.060**	-0.060	-0.060*	-0.060	-0.060
ñ	NADMEXP_EXP	0.015*	0.015**	0.020	0.020*	0.020	0.020	0.020	0.020	0.010	0.010
	PERSEXP_EXP	0.010	0.010	0.020	0.020	0.030	0.030*	0.040	0.040*	0.080	0.080***
	NFCL_TR_3M	0.400*	0.400***								
	NFCL_TR_6M			0.280	0.280***						
	NFCL_TR_9M					0.190	0.190***				
	NFCL_TR_12M							0.150	0.150***		
	NFCL_TR									0.070	0.070***
	LIQ	-0.280***	-0.280***	-0.470***	-0.470***	-0.600***	-0.600***	-0.650***	-0.650***	-0.720***	-0.720***
S	FSI	-0.020*	-0.020**	-0.040*	-0.040**	-0.050*	-0.050**	-0.060*	-0.060***	-0.080**	-0.080***
iablo	HHI_DEP	0.300*	0.300***	0.530**	0.530***	0.680**	0.680***	0.770**	0.770***	0.930**	0.930***
o var	CPI	-0.110	-0.110**	-0.220*	-0.220**	-0.310*	-0.310***	-0.360*	-0.360***	-0.490*	-0.490***
lacro	RDEP_GR	-0.050	-0.050*	-0.090*	-0.090**	-0.120*	-0.120**	-0.150*	-0.150**	-0.210**	-0.210**
≥	DC_100D_SHR	0.020***	0.020***	0.037***	0.037***	0.050***	0.050***	0.050***	0.050***	0.060***	0.060***
	NORM_RES	0.001***	0.001***	0.002**	0.002***	0.003**	0.003***	0.003**	0.003***	0.003**	0.003***
	Bank FE	Yes	Yes								
	Time FE	No	No								
	Heteroscedasticity	Two-way clusterization	HAC								
	Autocorrelation		HAC								
	Ν	1,656	1,656	1,656	1,656	1,656	1,656	1,656	1,656	1,656	1,656
	Within R <sup>2</sup>	0.989	0.771	0.967	0.754	0.947	0.747	0.928	0.746	0.887	0.728

 Table 14. Transmission to Deposit Rates vs Transmission to Credit Rates, 2018m1 – 2023m12

		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
	Transmission horizon	3m		6m		9m		12m		LR	
	NFCL_SHR	0.220*	0.220***	0.300	0.300***	0.310	0.310**	0.320	0.320**	0.170	0.170
	P_VK	0.003	0.003	0.004	0.004	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010
	INTREV_A	0.190	0.190	0.870	0.870	1.890*	1.890	2.640*	2.640*	5.140*	5.140*
	NPL_AGR	-0.050	-0.050***	-0.087*	-0.087***	-0.130*	-0.130***	-0.160*	-0.160***	-0.370**	-0.370***
	RES_A	0.020	0.020	0.050	0.050	0.090	0.090	0.120	0.120	0.290	0.290*
	FU_CST_HH	-0.110	-0.110*	-0.060	-0.060	-0.020	-0.020	-0.010	-0.010	0.040	0.040
	FU_CST_NFCL	0.150	0.150***	0.230	0.230***	0.210	0.210*	0.170	0.170	-0.090	-0.090
oles	VK_LIAB	0.049*	0.049***	0.050	0.050***	0.050	0.050*	0.050	0.050	0.080	0.080
arial	SHORTSEC_A	0.010	0.010	-0.010	-0.010	-0.030	-0.030	-0.040	-0.040	-0.020	-0.020
vel v	OVDP_A	0.040	0.040***	0.070	0.070***	0.080	0.080**	0.080	0.080*	0.030	0.030
k-lev	LIQ_A	-0.050	-0.050*	-0.090	-0.090*	-0.140	-0.140*	-0.170	-0.170*	-0.340	-0.340*
Ban	TERMDEP_LIAB	-0.010	-0.010	-0.020	-0.020	-0.030	-0.030	-0.030	-0.030	-0.010	-0.010
	NADMEXP_EXP	-0.010	-0.010*	-0.020	-0.020	-0.020	-0.020	-0.020	-0.020	-0.030	-0.030
	PERSEXP_EXP	0.020	0.020**	0.040	0.040**	0.050	0.050**	0.070	0.070**	0.120	0.120**
	HHD_TR_3M	0.270*	0.270***								
	HHD_TR_6M			0.200*	0.200***						
	HHD_TR_9M					0.150	0.150***				
	HHD_TR_12M							0.130	0.130**		
	HHD_TR									0.100	0.100
	LIQ	-0.010	-0.010	-0.030	-0.030	-0.050	-0.050	-0.060	-0.060	-0.0700	-0.070
bles	FSI	0.010	0.010	0.010	0.010	0.020	0.020	0.020	0.020	0.05	0.050
aria	HHI_CRED	0.130	0.130*	0.250	0.250**	0.360	0.360*	0.440	0.440*	0.520	0.520
cro v	CPI	0.030	0.030	0.030	0.030	0.030	0.030	0.020	0.020	-0.080	-0.080
Mag	RCRED_GR	0.048**	0.048***	0.084**	0.084***	0.100**	0.100**	0.120**	0.120**	0.180	0.180*
	DC_100D_SHR	0.007***	0.007*	0.016***	0.016*	0.024***	0.024*	0.030***	0.030*	0.055**	0.055*
	Bank FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
	Time FE	No	No	No	No	No	No	No	No	No	No
	Heteroscedasticity	Two-way clusterization	HAC	Two-way clusterization	HAC	Two-way clusterization	HAC	Two-way clusterization	HAC	Two-way clusterization	HAC
	Autocorrelation		HAC		HAC		HAC		HAC		HAC
	Ν	1,656	1,656	1,656	1,656	1,656	1,656	1,656	1,656	1,656	1,656
	Within R <sup>2</sup>	0.998	0.492	0.991	0.455	0.977	0.434	0.96	0.422	0.839	0.375

 $\textbf{Table 15.}\ Transmission\ to\ Credit\ Rates\ vs\ Transmission\ to\ Deposit\ Rates,\ 2018m1-2023m12$ 

		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
	Transmission horizon	Зm		6m		9m		12m		LR	
	HHD_SHR	0.280	0.282*	0.640	0.638*	0.950	0.947**	1.040	1.043*	0.760	0.760
	P_VK	-0.020	-0.019**	-0.030	-0.028**	-0.030	-0.031*	-0.030	-0.031*	-0.030	-0.030
	INTREV_A	-0.750	-0.753*	-1.240	-1.239*	-1.540	-1.542*	-1.720	-1.717*	-1.530	-1.530
	NPL_AGR	0.000	0.000	-0.010	-0.010	-0.010	-0.010	-0.020	-0.020	-0.030	-0.030
	RES_A	0.108*	0.108***	0.197*	0.197***	0.257**	0.257***	0.300**	0.300***	0.400**	0.400***
oles	FU_CST_HH	0.010	0.010	0.040	0.040	0.070	0.070	0.130	0.130	0.390	0.392*
arial	FU_CST_NFCL	-0.060	-0.060	-0.120	-0.120	-0.160	-0.160*	-0.180	-0.181*	-0.100	-0.100
/el v	VK_LIAB	0.010	0.010	0.010	0.010	0.020	0.020	0.020	0.020	0.020	0.020
k-lev	SHORTSEC_A	-0.030	-0.027**	-0.030	-0.030	-0.020	-0.020	-0.020	-0.020	-0.030	-0.030
Ban	LIQ_A	0.020	0.020	0.020	0.020	0.020	0.020	0.010	0.010	-0.080	-0.080
	TERMDEP_LIAB	0.000	0.000	0.010	0.010	0.020	0.020	0.020	0.020	0.030	0.030
	NADMEXP_EXP	0.010	0.013*	0.020	0.020	0.020	0.020	0.020	0.020	0.020	0.020
	PERSEXP_EXP	-0.005	-0.005	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010	0.030	0.030
	INTEXP_EXP	0.036*	0.036***	0.061*	0.061***	0.072*	0.072***	0.080*	0.080***	0.070	0.070**
	INTREV_REV	-0.000	-0.000	0.002	0.002	0.005	0.005	0.008	0.008	0.025*	0.025*
	LIQ	-0.260***	-0.260***	-0.400***	-0.400***	-0.470***	-0.470***	-0.500***	-0.500***	-0.480***	-0.480***
S	FSI	-0.020	-0.020*	-0.033*	-0.033*	-0.044*	-0.044**	-0.050*	-0.050**	-0.070**	-0.070***
iable	HHI_DEP	-2.940***	-2.940***	-5.120***	-5.120***	-6.530***	-6.530***	-7.280***	-7.280***	-8.330***	-8.330***
var	CPI	-0.170**	-0.170***	-0.340***	-0.340***	-0.470***	-0.470***	-0.550***	-0.550***	-0.790***	-0.790***
acro	RDEP_GR	-0.080**	-0.080***	-0.160***	-0.160***	-0.200***	-0.200***	-0.250***	-0.250***	-0.340***	-0.340***
Σ	DC_100D_SHR	0.002	0.002	-0.002	-0.002	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010
	NORM_RES	0.001**	0.001***	0.002**	0.002***	0.002**	0.002***	0.003**	0.003***	0.003*	0.003***
	Banks FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
	Time FE	No	No	No	No	No	No	No	No	No	No
	Heteroscedasticity	Two-way clusterization	HAC	Two-way clusterization	HAC	Two-way clusterization	HAC	Two-way clusterization	HAC	Two-way clusterization	HAC
	Autocorrelation		HAC		HAC		HAC		HAC		HAC
	Ν	1,656	1,656	1,656	1,656	1,656	1,656	1,656	1,656	1,656	1,656
	Within R <sup>2</sup>	0.987	0.747	0.959	0.749	0.936	0.754	0.916	0.758	0.901	0.75

# Table 16. Determinants of Transmission to Deposit Rates for Different Horizons, 2018m1 – 2023m12, (two largest banks excluded)

		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
	Transmission horizon	Зm		6m		9m		12m		LR	
	HHD_SHR	0.340	0.342**	0.750	0.751***	1.110	1.114***	1.320	1.317***	1.730	1.733***
	HHD_SHR_inter	0.010	0.010	0.010	0.010	0.010	0.010	0.020	0.020	0.050	0.050
	P_VK	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010
	INTREV_A	-0.760	-0.760*	-1.270	-1.270*	-1.590	-1.590*	-1.800	-1.800*	-1.760	-1.760
	NPL_AGR	0.000	0.000	-0.010	-0.010	-0.020	-0.020	-0.020	-0.020	-0.040	-0.040
S	RES_A	0.050	0.050**	0.110	0.110***	0.161*	0.161***	0.199*	0.199***	0.350**	0.350***
iable	FU_CST_HH	0.040	0.040	0.100	0.100	0.170	0.170	0.250	0.250	0.620	0.620**
l var	FU_CST_NFCL	-0.050	-0.050	-0.110	-0.110	-0.150	-0.150	-0.170	-0.170*	-0.170	-0.170
leve	VK_LIAB	-0.002	-0.002	-0.005	-0.005	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010
ank-	SHORTSEC_A	-0.020	-0.020*	-0.020	-0.020	-0.020	-0.020	-0.010	-0.010	-0.010	-0.010
â	LIQ_A	0.040	0.040	0.070	0.070	0.080	0.080	0.080	0.080	0.010	0.010
	TERMDEP_LIAB	-0.020	-0.020*	-0.020	-0.020	-0.020	-0.020	-0.010	-0.010	0.010	0.010
	NADMEXP_EXP	0.010	0.010*	0.020	0.020*	0.020	0.020	0.020	0.020	0.010	0.010
	PERSEXP_EXP	0.020	0.020*	0.030	0.030*	0.040	0.040*	0.040	0.040*	0.080	0.080**
	INTEXP_EXP	0.010	0.010	0.020	0.020	0.020	0.020	0.030	0.030	0.030	0.030
	INTREV_REV	0.002	0.002	0.005	0.005	0.010	0.010	0.010	0.010	0.031**	0.031**
	LIQ	-0.290***	-0.290***	-0.440***	-0.440***	-0.530***	-0.530***	-0.570***	-0.570***	-0.560***	-0.560***
	FSI	-0.030**	-0.030***	-0.050***	-0.050***	-0.070***	-0.070***	-0.080***	-0.080***	-0.100***	-0.100***
oles	HHI_DEP	0.390**	0.390***	0.610**	0.610***	0.720**	0.720***	0.780**	0.780***	0.750*	0.750***
arial	HHI_DEP_inter	0.021*	0.021***	0.035*	0.035***	0.044*	0.044***	0.047*	0.047***	0.047*	0.047***
cro v	CPI	-0.120	-0.120**	-0.200*	-0.250***	-0.350**	-0.350***	-0.400**	-0.400***	-0.590**	-0.590***
Mag	RDEP_GR	-0.040	-0.040	-0.090	-0.090*	-0.121*	-0.121**	-0.145*	-0.145**	-0.216**	-0.216**
	DC_100D_SHR	0.018**	0.018***	0.027**	0.027***	0.032*	0.032***	0.034*	0.034***	0.032*	0.032*
	NORM_RES	0.001**	0.001***	0.002**	0.002***	0.002*	0.002***	0.002*	0.002***	0.003*	0.003**
	Bank FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
	Time FE	No	No	No	No	No	No	No	No	No	No
	Heteroscedasticity	Two-way clusterization	HAC	Two-way clusterization	HAC	Two-way clusterization	HAC	Two-way clusterization	HAC	Two-way clusterization	HAC
	Autocorrelation		HAC		HAC		HAC		HAC		HAC
	Ν	1,800	1,800	1,800	1,800	1,800	1,800	1,800	1,800	1,800	1,800
	Within R <sup>2</sup>	0.987	0.715	0.963	0.716	0.943	0.72	0.922	0.723	0.877	0.701

 Table 17. Determinants of Transmission to Deposit Rates for Different Horizons, 2018m1 – 2023m12, (nonlinear effects of market concentration)

		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
	Transmission horizon	3m		6m		9m		12m		LR	
	NFCL_SHR	0.170	0.173**	0.300	0.301***	0.390	0.386***	0.450	0.451***	0.760	0.756**
	NFCL_SHR_inter	-0.010	-0.010	-0.020	-0.020	-0.030	-0.030	-0.030	-0.030	-0.060	-0.060
	P_VK	0.000	0.000	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010	-0.020	-0.020
	INTREV_A	0.240	0.240	0.710	0.710	1.470	1.470*	2.090	2.090**	4.470	4.470**
	NPL_AGR	-0.010	-0.010	-0.010	-0.010	-0.020	-0.020*	-0.030	-0.027*	-0.090	-0.090**
	RES_A	0.000	0.000	0.020	0.020	0.030	0.030	0.050	0.050	0.270	0.266**
bles	FU_CST_HH	-0.010	-0.010	0.020	0.020	0.020	0.020	0.000	0.000	-0.400	-0.396*
aria	FU_CST_NFCL	0.060	0.060**	0.040	0.040	-0.010	-0.010	-0.060	-0.060	-0.250	-0.250
vel v	VK_LIAB	0.004***	0.004**	0.006**	0.006*	0.007*	0.007*	0.010	0.010	0.010	0.010
k-lev	SHORTSEC_A	-0.010	-0.010**	-0.020	-0.018*	-0.020	-0.020	-0.020	-0.020	0.030	0.030
Ban	OVDP_A	0.010	0.010**	0.030	0.030***	0.040	0.040***	0.050	0.050***	0.060	0.057*
	LIQ_A	0.010	0.010	0.020	0.020	0.020	0.020	0.030	0.030	0.060	0.060
	TERMDEP_LIAB	-0.030*	-0.030***	-0.050*	-0.050***	-0.050	-0.050***	-0.060	-0.060***	-0.110	-0.100***
	NADMEXP_EXP	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010	0.000	0.000
	PERSEXP_EXP	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.010	0.010
	INTEXP_EXP	0.010	0.011**	0.010	0.010	0.010	0.010	0.020	0.020	0.020	0.020
	INTREV_REV	-0.003	-0.003	-0.005	-0.005	-0.010	-0.010	-0.010	-0.010	-0.010	-0.010
	LIQ_5	-0.090***	-0.090***	-0.140**	-0.140***	-0.170**	-0.170***	-0.200**	-0.200***	-0.300*	-0.300***
S	FSI	-0.000	-0.000	0.004	0.004	0.010	0.010	0.010	0.010	0.050*	0.050
iable	HHI_CRED	0.156**	0.156***	0.268**	0.268***	0.376**	0.376***	0.467**	0.467***	1.020*	1.020***
var	HHI_CRED_inter	0.002	0.002	0.002	0.002	0.001	0.001	-0.000	-0.000	-0.010	-0.010
acro	CPI	0.002	0.002	-0.010	-0.010	-0.020	-0.020	-0.030	-0.030	-0.170	-0.170
Σ	RCRED_GR	0.026***	0.026**	0.048**	0.048**	0.068**	0.068**	0.084**	0.084**	0.170*	0.170**
	DC_100D_SHR	0.007***	0.007**	0.012**	0.012**	0.018**	0.018**	0.023**	0.023**	0.061*	0.060***
	Banks FE	Yes	Yes								
	Time FE	No	No								
	Heteroscedasticity	Two-way clusterization	HAC								
	Autocorrelation		HAC								
	Ν	3,960	3,960	3,960	3,960	3,960	3,960	3,960	3,960	3,960	3,960
	Within R <sup>2</sup>	0.997	0.293	0.992	0.303	0.983	0.303	0.973	0.302	0.802	0.281

 Table 18. Determinants of Transmission to Credit Rates for Different Horizons, 2018m1 – 2023m12 (nonlinear effects of market concentration)

	(1)	(2)	(3)	(4)					
	LR								
LIQ_5	-1.055***	-1.057***	-1.055***	-1.057***					
FSI	0.104*	0.020	0.104***	0.020					
HHI_DEP	1.572***	1.487***	1.572***	1.487***					
RDEP_GR	0.090	-0.020	0.090	-0.020					
DC_100D_SHR	0.126***	0.077***	0.126***	0.077***					
NORM_RES	0.000	0.000	0.000	0.002***					
CPI	0.080	-0.437*	0.080	-0.437***					
CPI <sup>2</sup>		0.724**		0.724***					
Bank FE	Yes	Yes	Yes	Yes					
Time FE	No	No	No	No					
Heteroscedasticity	Two-way clusterization	Two-way clusterization	HAC	HAC					
Autocorrelation			HAC	HAC					
Ν	2,659	2,659	2,659	2,659					
Within R <sup>2</sup>	0.71	0.72	0.36	0.37					

# Table 19. Determinants of Transmission to Deposit Rates, 2015m1 – 2023m12

	(1)	(2)	(3)	(4)					
	LR								
LIQ_5	-0.297	-0.327*	-0.297***	-0.327***					
FSI	0.112***	0.037	0.112***	0.037					
HHI_CRED	0.813***	0.599**	0.813***	0.599***					
RCRED_GR	0.092*	0.125**	0.093***	0.125***					
DC_100D_SHR	0.084***	0.042	0.084***	0.042**					
CPI	-0.059	-0.424**	-0.059	-0.424***					
CPI <sup>2</sup>		0.759***		0.759***					
Banks FE	Yes	Yes	Yes	Yes					
Time FE	No	No	No	No					
Heteroscedasticity	Two-way clusterization	Two-way clusterization	HAC	HAC					
Autocorrelation			HAC	HAC					
Ν	5,807	5,807	5,807	5,807					
Within R <sup>2</sup>	0.726	0.729	0.147	0.156					

 Table 20. Determinants of Transmission to Credit Rates, 2015m1 – 2023m12



# **APPENDIX B. FIGURES**

**Figure 7.** Asymmetry of Transmission to Deposit Rates (a negative sign means the response to a decrease in the market rate is stronger)



Nonsignificant
 Significant at 10%

Figure 8. Pass-Through Asymmetry for Deposits and Share of Deposits



Figure 9. Histogram Plot of the Long-Run Pass-Through for NFC Loans



Figure 10. Long-Run Pass-Through for NFC Loans



Figure 11. Long-Run Pass-Through for NFC Loans and Share of NFC Loans



Figure 12. Asymmetry of Transmission to Credit Rates (a negative sign means the response to a decrease in the market rate is stronger), transmission asymmetry, bank code



Nonsignificant 
 Significant at 10%

Figure 13. Long-Run Pass-Through for NFC Loans and Share of Loans



Figure 14. Impulse Response Functions for State, Foreign, and Domestic Private Banks, %, month



Figure 15. Time Variant Pass-Through for Deposit Rates, long-run<sup>15</sup>



Figure 16. Time Variant Pass-Through for NFC Loan Rates, long-run



Figure 17. Time-Varying Pass-Through for (A) Deposit Rates and (B) Credit Rates on Different Horizons





Figure 18. Kernel Densities of Pass-Through Strength on Different Horizons



Figure 19. Time-Varying Transmission Strength of the Banks that Changed Ownership Status, (A) – deposit rates, (B) – credit rates

Note: Dots on the plots mark months when ownership status has been changed. The median is estimated for the subsample excluding banks with ownership changes. Confidence intervals were estimated as +/- 1.96\*s.e.



Figure 19 (continued). Time-Varying Transmission Strength of the Banks that Changed Ownership Status, (A) – deposit rates, (B) – credit rates



Figure 20. Transmission to Deposit Rates: Parameters of Baseline Specification for Different Groups of Banks



Figure 21. Transmission to Credit Rates: Parameters of Baseline Specification for Different Groups of Banks